# Hourly temperature data do not support the views of the Climate Deniers: Evidence from Barrow Alaska

Kevin F. Forbes $^{1,1}$ 

<sup>1</sup>Energy and Environmental Data Science

December 1, 2022

#### Abstract

Survey evidence has indicated that a significant percentage of the population does not fully embrace the scientific consensus regarding climate change. This paper assesses whether the hourly temperature data support this denial. The analysis examines the relationship between hourly CO2 concentration levels and temperature using hourly data from the NOAA-operated Barrow observatory in Alaska. At this observatory, the average annual temperature over the 2015-2020 period was about 3.37 oC higher than in 1985–1990. A time-series model to explain hourly temperature is formulated using the following explanatory variables: the hourly level of total downward solar irradiance, the CO2 value lagged by one hour, proxies for the diurnal variation in temperature, proxies for the seasonal temperature variation, and proxies for possible non-anthropomorphic drivers of temperature. The purpose of the time-series approach is to capture the data's heteroskedastic and autoregressive nature, which would otherwise "mask" CO2's "signal" in the data. The model is estimated using hourly data from 1985 through 2015. The results are consistent with the hypothesis that increases in CO2 concentration levels have nontrivial consequences for hourly temperature. The estimated annual contributions of factors exclusive of CO2 and downward total solar irradiance are very small. The model was evaluated using out-of-sample hourly data from 1 Jan 2016 through 31 Aug 2017. The model's out-of-sample hourly temperature predictions are highly accurate, but this accuracy is significantly degraded if the estimated CO2 effects are ignored. In short, the results are consistent with the scientific consensus on climate change.

#### Hourly temperature data do not support the views of the Climate Deniers: Evidence from Barrow Alaska

Kevin F. Forbes, Ph.D<sup>1</sup>

<sup>1</sup> Energy and Environmental Data Science, Malahide, Ireland

Corresponding author: Kevin F. Forbes (Kevin.F.Forbes@EEDS.Solutions)

Key Points:

1) At NOAA's Barrow Observatory in Alaska, the annual temperature during 2015-2020 was about 3.37 °C higher than in 1985-1990. 2) Virtually all the upward trend in annual temperature through 2015 can be attributed to higher  $CO_2$  concentrations. 3) The model's out-of-sample predictions are more accurate if the estimated associations between  $CO_2$  and temperature are not ignored.

#### Abstract

Survey evidence has indicated that a significant percentage of the population does not fully embrace the scientific consensus regarding climate change. This paper assesses whether the hourly temperature data support this denial. The analysis examines the relationship between hourly  $CO_2$  concentration levels and temperature using hourly data from the NOAA-operated Barrow observatory in Alaska. At this observatory, the average annual temperature over the 2015-2020 period was about 3.37 °C higher than in 1985–1990.

A time-series model to explain hourly temperature is formulated using the following explanatory variables: the hourly level of total downward solar irradiance, the  $CO_2$  value lagged by one hour, proxies for the diurnal variation in temperature, proxies for the seasonal temperature variation, and proxies for possible non-anthropomorphic drivers of temperature. The purpose of the time-series approach is to capture the data's heteroskedastic and autoregressive nature, which would otherwise "mask"  $CO_2$ 's "signal" in the data. The model is estimated using hourly data from 1985 through 2015. The results are consistent with the hypothesis that increases in  $CO_2$  concentration levels have nontrivial consequences for hourly temperature. The estimated annual contributions of factors exclusive of  $CO_2$  and downward total solar irradiance are very small. The model was evaluated using out-of-sample hourly data from 1 Jan 2016 through 31 Aug 2017. The model's out-of-sample hourly temperature predictions are highly accurate, but this accuracy is significantly degraded if the estimated  $CO_2$  effects are ignored. In short, the results are consistent with the scientific consensus on climate change.

#### Plain Language Summary

According to the IPCC and other scientific organizations, "it is extremely likely that human influence has been the dominant cause of the observed increase in global temperatures since the mid-20th century." However, a significant percentage of the population does not fully embrace this consensus. Using data from the Barrow Atmospheric Observatory, this paper assesses whether the hourly temperature data support this apparent denial. It is first noted that the average annual temperature at Barrow over the 2015-2020 period was about  $3.37^{\circ}$ C higher than in the 1985-1990 period. The formal analysis employs hourly solar irradiance, CO<sub>2</sub>, and temperature data. The model controls for possible non-anthropomorphic drivers of annual temperature and other factors. The model was estimated using hourly data over the time interval 1 Jan 1985 through 31 Dec 2015. The estimated annual effects of CO<sub>2</sub> are significant in magnitude, while the non-anthropomorphic drivers exclusive of solar irradiance are quantitively unimportant. The model is evaluated over the 1 Jan 2016 through 31 Aug 2017 time interval. The model's out-of-sample hourly temperature predictions are highly accurate, but this accuracy is degraded if the estimated CO<sub>2</sub> effects are ignored. In short, the results are consistent with the scientific consensus on climate change.

#### Index Terms

6620 Science Policy

1630 Impacts of Global Change

1616 Climate Variability

9315 Arctic Region

3270 Time series analysis

1986 Statistical methods: Inferential

#### Key Words:

 $\rm CO_2$  Concentrations, Hourly Temperature, Downward total solar irradiance, Climate Change, Arctic Region, Alaska

**Acronyms:** AMAP, Arctic Monitoring and Assessment Program, ARCH, Autoregressive conditional heteroskedasticity; ARMA, autoregressive–moving-average; ARMAX, autoregressive–moving-average with exogenous inputs; ECMWF, European Centre for Medium-Range Weather Forecasts. MFP, multivariable fractional polynomial; RMSE, root-mean-squared-error.

#### 1. Introduction

According to the IPCC, "It is extremely likely that human influence has been the dominant cause of the observed increase in global temperatures since the mid-20th century "(IPCC, 2013, p. 17). As early as 2001, the science academies of Australia, Belgium, Brazil, Canada, the Caribbean, China, France, Germany, India, Indonesia, Ireland, Italy, Malaysia, New Zealand, Sweden, Turkey, and the United Kingdom all endorsed the IPCC's Third Assessment (*Australian Academy of Sciences et al., 2001*). A more recent list of scientific academies that have accepted this view includes the science academies in Japan, Russia, and the USA. (National Academies of Science, 2005). These institutes are not indicating that human activity is only partly responsible for climate change. Instead, they have indicated that human activity is the dominant driver.

In the United States, a country in which a nontrivial number of climate deniers hold powerful elected positions, a group of 18 highly respected scientific organizations explicitly endorsed the scientific consensus on climate change in a 2009 letter to U.S. policymakers (American Association for the Advancement of Science, 2009). This Letter was released again in 2016 by a larger group of 31 scientific organizations (American Association for the Advancement of Science, 2016). The updated Letter makes the following point:

"Observations throughout the world make it clear that climate change is occurring, and rigorous scientific research concludes that the greenhouse gases emitted by human activities are the primary driver. This conclusion is based on multiple independent lines of evidence and the vast body of peer-reviewed science." AAAS, 2016

This paper's starting point is the observation that the survey data does not fully reflect the scientific consensus. This paper applies methods developed to address issues in economics and finance to assess whether the temperature data at the Barrow Atmospheric Observatory in northern Alaska supports this view. While some might sharply question the approach employed in this paper because the methodology is "unorthodox" relative to the conventional meteorological framework, it may be worth noting that the methodology applied in this paper has revolutionized the analysis in other sectors when the data are found to be autoregressive and heteroskedastic in nature. One modest example of this is Forbes and Zampelli (2019), who analyzed  $CO_2$ emissions from the Irish power grid using the methods presented in this paper after observing that the emission levels had autoregressive and heteroskedastic properties. These properties will be shown to be highly relevant when modeling hourly temperature. Ignoring these properties makes extracting  $CO_2$ 's "signal" from the "noisy" data almost impossible.

In terms of organization, section 2 of the paper discusses the survey data. Section 3 summarizes the views of individuals identified as being climate deniers within the scientific community. Section 4 discusses the data used in the analysis. To provide context, the trends in hourly temperature, downward total solar irradiance, and  $CO_2$  concentrations at the Barrow Atmospheric Observatory are reported. In response to an assertion about a lack of recent warming relative to the pre-1940 period by Lindzen (2020, pp. 12-13), the annual temperature at the nearby Barrow Airport from 1921 through 2020 is reported. The time-series nature of hourly temperature at Barrow is also discussed to facilitate the modeling discussion in the remaining sections of the paper. Section 5 introduces a modeling framework to examine the possible association between  $CO_2$  concentrations and hourly temperature. Section 6 discusses the estimation process and also presents the results. Section 7 evaluates the model. The paper's findings are discussed in section 8.

#### 2. The Survey Evidence

A 2019 YouGov survey of 30,000 individuals that are believed to be representative of the online population in 28 countries indicated that there were only 14 countries in which 50 % or more of the respondents would agree with the statement that "The climate is changing and human activity is mainly responsible" (Figure 1). A significant number of the respondents indicated that human activity is only partly responsible for climate change. For example, while 40% of the respondents in Denmark agreed with the scientific consensus, 48% agreed with the view that "...human activity is partly responsible, **together with other factors** (**emphasis added**). In the United Kingdom, 51% endorsed the scientific consensus, while 37% believe that human activity is only partly responsible. In China, 45% endorsed the scientific consensus, while 48% believe human activity is only partly responsible. In the USA, 38% endorsed the scientific consensus, 37% reported that they believe that human activity is only partly responsible for climate change, 9% believe that human activity is not a driver of climate change, and 6% reported that they do not believe that the climate is changing.

India	71		23
Thailand	69		27
Spain	69		27
Indonesia	69		24
Italy	66	29	
Vietnam	64	32	
Philippines	62	31	4
Singapore	54	39	
Tawain	53	42	
Qater	52	41	
Kuwait	52	34	4
UAE	52	33	6
Great Britain	51	37	
Hong Kong	50	45	
Finland	49	38	5
Germany	49	36	5
France	48	37	4
Malaysia	48	43	
Bahrain	46	41	4
China	45	48	
Australia	44	43	55
Oman	43	47	
Egypt	42	38	7
Denmark	40	48	4
USA	38	37	9 6
Sweden	36	48	6
Saudi Arabia	35 3	6	7 5
Norway	35 4	8	8

The climate is changing and human activity is mainly responsible The climate is changing and human activity is partly responsible, together with other factors The climate is changing and human activity is not responsible at all (%) The climate is not changing

Created with Datawrapper

#### Source: Source: 10.5281/zenodo.5833580

# Figure 1. Responses to a 2019 YouGov survey question posed to 30,000 people in 28 countries. Thinking about the global environment... In general, which of the following statements, if any, best describes your view?"

While it is tempting to attribute the findings for China in Figure 1 as evidence of a form of climate denial by a large proportion of its population, the recent findings by Yang et al. (2021) would seem to suggest that a sincere misunderstanding of the nature of climate change might be a more important consideration. In other countries, other survey data are largely consistent with the data presented in Figure 1. For example, in a 2019

Irish Times/Iposos MRBI poll (Leahy, P., 2019), respondents were asked if they agreed with the following statement: "I don't think climate change will be as bad as some say so I'm not that worried about it." While 57% of the respondents implicitly endorsed the scientific consensus by disagreeing with the statement, 33% agreed. In this same poll, only 44% of the respondents agreed with the statement, "I am okay with the price of oil, gas, petrol and diesel increasing to help tackle climate change." This is obviously not a majority and thus represents a challenge to implementing policies to reduce emissions.

A November 2018 survey of 1,202 adults by the Energy Policy Institute at the University of Chicago and the AP-NORC Center yields useful insights (EPIC, 2018). According to this survey, 57% of the respondents were willing to pay a \$1 monthly fee to combat climate change. About 23% were willing to pay 40 USD per month. However, 43 percent were unwilling to pay anything, highlighting the challenge of doing anything significant to reduce emissions. Acceptance of the view that human activity contributes to climate change was a useful indicator of whether respondents were willing to pay to reduce emissions.

Suggestive of the possible political implications of the polling data, the UNFCCC secretariat (United Nations Framework Convention on Climate Change) issued a report in September 2021 that indicated that the combined updated Paris Accord pledges fall short of what it will take to meet the goals of the Paris Accords. Specifically, even with the updated pledges, projected GHG emissions in 2030 are only about 0.5% lower than in 2010, which is far lower than what it would take to limit global warming to below two °C (UNFCCC Secretariat, 2021a). The COP26 meetings that were held in November of 2021 have done little to improve the prospects that the goals of the 2015 Paris Accords will be met. The United States did announce its good intentions, but climate deniers will most likely make those goals very difficult to achieve. The conference faced other challenges including objections to phasing out coal. While the conference made progress in the areas of carbon markets and finance, the fact remains that there is a significant emissions gap (UNFCCC Secretariat, 2021b).

#### **3** The Views of the Climate Deniers from within the Scientific Community

Somewhat surprisingly, some prominent individuals from within the scientific community who have been labeled as climate deniers have actually conceded that increases in CO2 concentrations have consequences for surface warming. For example, the CO<sub>2</sub> Coalition (2015), a sharp critic of the scientific consensus, whose members include the well-known influencers Richard Lindzen, Patrick Michaels, Roy Spence, and William Happer, has explicitly acknowledged the greenhouse effect. It notes that predicting greenhouseinduced warming is difficult because atmospheric processes are very complicated. It then pivots back and reports that it believes that the data suggests that the warming associated with a doubling of  $CO_2$  levels will be very modest. In its words,

"Basic physics implies that more atmospheric  $CO_2$  will increase greenhouse warming. However, atmospheric processes are so complicated that the amount of warming cannot be reliably predicted from first principles. Recent observations of the atmosphere and oceans, together with geological history, point to very modest warming, about 1 C (1.8 F) if atmospheric  $CO_2$  levels are doubled."  $CO_2$  Coalition, 2015

The  $CO_2$  Coalition's assertion that the warming associated with a doubling of  $CO_2$  will be modest appears to be largely premised on a belief that the recent warming is about the same as before the 1940s (Lindzen, 2020, pp. 12-13). As will be seen, this belief is not supported by the data in northern Alaska.

#### 4 An Overview of the Changing Climate in Northern Alaska

The study employs temperature, solar radiation, and  $CO_2$  data reported by the Barrow (BRW) Atmospheric Observatory. This is one of the baseline observatories of the Earth System Research Laboratory (ESRL), Global Monitoring Division (GMD), of the National Oceanic and Atmospheric Administration (NOAA). It is located near sea level about 8 km east of Utqiagivik (formerly Barrow), Alaska at 71.3230 degrees north and 256.6114 degrees West (Vasel et al., 2020). Continuous atmospheric measurements of  $CO_2$  have been recorded at this observatory since July 1973 (Thoning et al., 2021). Herbert et al. (1986) discuss how the data are processed. Peterson et al. (1986) discuss the first ten years (1973-1982) of operations and report

consistency of the Barrow results with the reported data from four neighboring locations. Tans and Thoning (2020) provide a general overview of the methods used to collect and process the  $CO_2$  data at Mauna Loa, one of NOAA's other baseline observatories. Along with the hourly temperature data corresponding to BRW, the  $CO_2$  data for BRW were downloaded using the following link: (http://www.esrl.noaa.gov/gmd/dv/data/).

Measurements of downward total solar irradiance have been reported at the BRW observatory since January 1976. Before 1998, the data were reported at three minutes intervals. The data were subsequently reported at one-minute intervals. For this study, the reported values were rolled up to hourly averages. Data were dropped from the analysis if the number of valid minutes of data for an hour was less than 15.

Consideration was given to the inclusion of  $CH_4$  data in the analysis. This action would have resulted in the loss of 26,381 hourly observations due to unavailable or invalid  $CH_4$  measurements. (the collection of the  $CH_4$  data commenced in 1986 but was subsequently suspended for about nine months in 2012/2013). The probable effect of this data loss on model convergence was an important consideration in excluding this variable from the analysis, model convergence being one of the major challenges of the methodology employed in this paper (STATA, 2021, p. 33). The omission of  $CH_4$  and other variables reflecting greenhouse gas concentrations represents a shortcoming in the analysis.

The sample for this study spans from 1 Jan 1985 through 31 Dec 2015. Data before 1 Jan 1985 were not employed in this study because the reported downward total solar irradiance data largely did not meet ESRL's standards before that date. For example, only about 31% of the downward total solar irradiance values in 1984 were deemed by ESRL to be valid. The 1 Jan 2016 - 31 Aug 2017 time interval is reserved for out-of-sample analysis. The evaluation period terminates on 31 Aug 2017 because of a significant data availability issue.

In thinking about meteorological issues at BRW, it is useful to begin by first noting the extremes and high level of variability in the level of downward total solar irradiance at this location. In terms of variability, the data from 2014 is instructive (Figure 2). Concerning the extremes, there are about 67 days of virtually total darkness each year ( about 18 Nov to 22 Jan), while the sun does not completely set from 11 May to 31 Jul.



Figure 2. The level of hourly downward total solar irradiance at BRW, 1 Jan 2014 – 31 Dec 2014

The average annual temperature at BRW has increased significantly since 1985 (Figure 3). Specifically, the average annual temperature over the 2015-2020 time period was about 3.37 °C higher than in 1985-1990. The temperature data reported by the PABR weather station at the nearby Barrow Airport from 1985 through 2020 are consistent with the trend at BRW (Figure 4). The PABR data also indicates that the four warmest years since 1921 occurred in 2016, 2017, 2018, and 2019. In these four years, the average annual temperature was about 5.03° C higher than the average annual temperature from 1921 through 1939. These findings do not support the assertion by Lindzen that the recent warming is about the same as before the 1940s (2020, pp. 12-13). In terms of the magnitudes of the recent warming, the increases are consistent with Arctic amplification, as explained by Pithan & Mauritsen (2014) and Winton (2006).

The upward trend in temperature at both BRW and PABR is consistent with the temperature trend for the Arctic noted by Post et al. (2019), Markon et al. (2018, p 1190-1192), and Thoman et al. (2020, p. 4). Box et al. (2019) have reported significant changes in nine key measures of the Arctic climate system over 1971 through 2017. The qualitative story is clear: "the transformation of the Arctic to a warmer, less frozen, and biologically changed region is well underway." (Thoman et al., 2020, p. 1). Consistent with these changes, the annual mean permafrost temperatures have increased at many locations throughout the Arctic (Romanovsky et al., 2017, p. 69). For example, based on data reported by EPA, the average annual permafrost temperature at the Deadhorse Permafrost Observatory ( https://permafrost.gi.alaska.edu) over the years 2015 through 2020 was about 2.81 °C higher than during the years 1985 through 1990 (EPA, 2021). In four of the 11 permafrost observatories whose 2020 annual temperatures are reported by EPA, the 2020 average temperatures were between -1 and 0°C. There is evidence that thawing has adverse implications for carbon emissions because of stimulated microbial decomposition (Schuur et al., 2021).

According to AMAP, "Arctic warming can also have effects far beyond the region: for example, the recent rapid warming of the Arctic appears to have created conditions favoring a persistent pattern in the jet stream that provokes unusual extreme temperature events in the Northern Hemisphere." (AMAP, 2019, p. 4). Taylor et al. (2017, p. 303) have indicated it is very likely that human activities have contributed to these trends. While the literature supports this finding, it has also been suggested that the significant natural weather and climate variability in the Arctic poses an attribution challenge (Taylor et al., 2017, p. 319). Consistent with this reported variability, both downward total solar irradiance and temperature at the hourly level are highly variable (Figures 5 and 6). Concerning the hourly  $CO_2$  concentration levels, there is a significant upward trend in the hourly  $CO_2$  concentration levels over the sample (Figure 7). Despite the upward trend in both  $CO_2$ concentrations and temperature, there is no visually obvious relationship between the two variables (Figure 8). While some climate deniers may be tempted to claim that the data in this figure vindicates their position, the view here is that a lack of correlation between two variables only rules out causality when the hypothesized relationship is quite simple.



Figure 3. The average hourly temperature at the Barrow Observatory, 1985 -2020



Figure 4. The average annual temperature at the PABR/Barrow Airport weather station, 1921 -2020



Figure 5. The hourly temperature at the Barrow Observatory, 1 Jan 1985 – 31 Dec 2016



Figure 6. Hourly downward total solar irradiance levels at the Barrow Observatory, 1 Jan 1985 – 31 Dec2016



Figure 7. Hourly CO<sub>2</sub> concentration levels at the Barrow Observatory, 1985 -2019



Figure 8. A scatter diagram of hourly temperature and  $CO_2$  concentration levels at BRW, 1 Jan 1985 – 31 Dec 2015

The autocorrelative nature of hourly temperature is an important characteristic of the data (Figure 9). As the figure indicates, the magnitude and the duration of the autocorrelative process are significant. In terms of magnitude, the estimated one-hour autocorrelation in temperature equals 0.9970, a value that is so large

that it is reasonable to wonder if there is a unit root issue. If this is indeed the case, the results of this study could be spurious for the reasons explained by Kennedy (2008, p. 301).

Fortunately, an Augmented Dickey-Fuller test yields a P-value that is less than 0.0001 both with and without a possible trend, and thus the null hypothesis of a unit root is rejected. Consistent with this finding, the Phillips-Perron test for a unit root also yields a P-value less than 0.0001 both with and without a possible trend. Consideration was given to further unit root testing using the DF-GLS test developed by Elliot et al. (1996). This test is regarded as a leading "second-generation" unit root test that avoids some of the shortcomings of the Augmented Dickey-Fuller and Phillips-Perron tests (Baum and Hurn, 2021, pp. 117-120 ). The application of this methodology requires a data series without any gaps. The Barrow data set has 325 gaps in terms of temperature, and thus, the DF-GLS test cannot be applied.

Fortunately, hourly temperature data analysis at another observatory in the polar region may be instructive. One of the few stations in the polar region that substantially meets the zero data gap requirements of the DF-GLS test is the Syowa station on East Ongle Island, located about 4km from the Antarctic continent with a latitude 69.0125° South and a longitude of 39.5900° East. This station is supported by the National Institute of Polar Research in Japan. The data from this station was obtained from NASA's CERES/ARM Validation Experiment (https://ceres-tool.larc.nasa.gov/ord-tool/jsp/SYN1degEd41Selection.jsp).

From 14 Apr 2002 through 31 Jan 2016, a period with 120,982 hours and no data gaps, the mean temperature at the Syowa Observatory was about -10.7 °C, with the hourly values ranging from 41.25 °C to 7.65 °C. At one hour lagged, the autocorrelation in temperature equals 0.9959, a value seemingly suggestive of a unit root issue. This possible suspicion is not supported by the Augmented Dickey-Fuller, Phillips–Perron, or the DF-GLS tests.

While the available tests do not support the null hypothesis of a unit root in the hourly temperature data, a quantitative analysis of hourly time-series temperature data needs to control its time-series nature to effectively extract the signal from the noise in the data. The method of ordinary least squares is woefully deficient in this regard. This point is consistent with a warning by Granger and Newbold (1974, p. 117), who note the following: "In our opinion the econometrician can no longer ignore the time series properties of the variables with which he is concerned - except at his [ or her ] peril." The consequences of ignoring their warning include inefficient estimates of the regression coefficients, suboptimal forecasts, and invalid tests of statistical significance. Unfortunately, an inspection of "Statistical Methods in the Atmospheric Sciences," authored by Wilks (2019), suggests that this warning has not been fully heeded in the atmospheric sciences.



Figure 9. The autocorrelations in hourly temperature at Barrow, 1 Jan 1985 – 31 Dec 2015

#### 5 An ARCH/ARMAX Model of Hourly Temperature

The model employed in this paper is an Autoregressive Conditional Heteroskedasticity/ Autoregressive– Moving-Average with Exogenous Inputs model of temperature (henceforth, an ARCH/ARMAX model of temperature). The ARCH terms are employed to model the conditional heteroskedasticity, an important consideration in the convergence process. The Autoregressive–Moving-Average (ARMA) component models the autocorrelations in temperature depicted in Figure 9. In this section, the role of the exogenous inputs is discussed.

Following from Forbes and St. Cyr (2017, 2019) and Forbes and Zampelli (2019, 2020), the modeling approach employed in this paper accepts the proposition that "All models are wrong; some models are useful" (Box et al., 2005, p. 440). They are all "wrong" because they represent a simplification of reality; they can be useful if important features of that reality are captured. A possibly related proposition that may be relevant during these times of sharp differences in opinions is "that all modeling results can easily be dismissed out of hand as being wrong, even if they are useful." In the case of this research, it may be asserted that the results are "wrong" because the model is adversely affected by "specification errors," "multicollinearity," "autocorrelation," "heteroskedasticity," "overfitting," and "unit-root issues." Other readers may conclude that the model is "wrong" because it somehow "forces" the estimated relationship between  $CO_2$  concentrations and temperature to be positive because both are rising over time (note: the correlation between temperature and  $CO_2$  equals -0.1495). Still, others will argue that the results are "biased" because the model's dependent variable is the natural logarithm of temperature.

Following from Forbes and Zampelli (2020, p. 13), this paper accepts the proposition that the "...vulnerability of a model to be deemed as wrong even though all models are "wrong" represents a challenge to the recognition of insights provided by models that are useful." Fortunately, this challenge can be addressed by assessing a model's predictive accuracy. Common sense informs us that a model that yields accurate predictions is useful if the evaluation interval is sufficiently long. Based on this perspective, the approach in this paper proceeds by estimating the model using 228,085 observations and performing an out-of-sample analysis with 13,175 observations.

In the model, the association between  $CO_2$  concentrations and temperature is presumed to be conditional on the level of downward total solar irradiance measured at the Earth's surface, downward total solar irradiance being the primary driver of the weather and climate system. The other drivers of the surface energy balance, such as upward and downward longwave irradiance, are not included as explanatory variables in the model because they are hypothesized to be affected by  $CO_2$  concentrations. Upward short-wave irradiance is not hypothesized to be directly affected by  $CO_2$  concentrations. Its inclusion as an explanatory variable is open to question, given that it is largely driven by downward solar irradiance and temperature. The inclusion of this variable would significantly reduce the sample size, given that ESRL only commenced reporting this variable in 1993.

In the model,  $CO_2$  concentrations are lagged one hour to avoid the issue of possible two-way causality between temperature and CO<sub>2</sub> concentrations. The model also includes binary variables representing the solar zenith angle, the hour-of-the-day, day-of-the-year, and year. These variables are included as proxies for the drivers of the diurnal variation in temperature, the seasonal variation in temperature, and the possible non-anthropomorphic drivers of temperature unrelated to total downward solar irradiance. In terms of functional form, linearity is not presumed. Instead, the data are permitted to speak for themselves on this important issue.

The initial version of the model is given by:

$$\begin{aligned} \ln \text{Temp}_{t} &= \alpha_{0} + \alpha_{1} \text{ZeroSolar}_{t} + \alpha_{2} \text{ Solar}_{t} + \alpha_{3} (\text{CO2}_{t-1} \text{*} \text{ZeroSolar}_{t}) \\ &+ \alpha_{4} (\text{CO2}_{t-1} \text{*} \text{Solar}_{t}) + \alpha_{5} \text{Solar}_{t} \text{*} \text{CO2}_{t-1} + \sum_{h=1}^{9} \beta_{h} \text{Angle}_{h} \\ &+ \sum_{i=2}^{24} \phi_{i} \text{HourofDay}_{i} + \sum_{j=2}^{365} \gamma_{j} \text{DOY}_{j} + \sum_{k=1985}^{2014} \delta_{k} \text{Year}_{k} (1) \end{aligned}$$
Where

Where

lnTemp<sub>t</sub> is the natural logarithm of temperature measured in Kelvin in hour t.

ZeroSolar<sub>t</sub> is a binary variable. The variable is assigned a value of one if the downward total solar irradiance level at Barrow in period t equals zero. Its value equals zero otherwise.

Solar<sub>t</sub> equals the downward total solar irradiance level at Barrow in period t.

 $CO2_{t-1}$  is the atmospheric level of  $CO_2$  concentrations at Barrow in hour t-1.

PosSolart is a binary variable that equals one if the level of downward total solar irradiance at Barrow in period t is positive. Its value equals zero otherwise.

Angle<sub>h</sub> is a vector of nine variables representing the solar zenith angle.

 $HourofDay_i$  is a series of 23 variables representing the hour of the day.

DOY<sub>i</sub> is a series of 364 binary variables representing the day of the year.

 $Year_k$  is a series of 30 binary variables representing the year.

Please note that  $\alpha_1$ ,  $\alpha_2$ , and  $\alpha_3$ , etc. are the coefficients corresponding to this linear version of the model. From (1), the total number of coefficients to be estimated equals 432. Some may strongly suspect that this number of explanatory variables indicates that the model is "overfitted." If this claim is true, the model would be unlikely to yield accurate out-of-sample predictions even if the within-sample explanatory power is very high (Brooks, 2019, p. 271). The "rule of thumb" by Trout (2006) that overfitting is avoided when there are at least ten observations per estimated coefficient does not support this possible suspicion given that the structural model present in this paper entails over 500 observations per estimated coefficient. Moreover, as will be seen, the model does not suffer from the consequences of overfitting in terms of out-of-sample predictive accuracy.

#### 6 Estimation and Results

The model was estimated using hourly data over the 1 Jan 1985 - 31 Dec 2015 time interval. The analysis was conducted in two distinct stages. In the first stage, the functional form given by Eq. (1) was evaluated. A nonlinear functional form was subsequentially identified.

The analysis also recognizes that the disturbance term's variance in a regression equation is heteroskedastic instead of homoscedastic, i.e., variable instead of constant over time. As suggested in the previous section, the accepted approach involves estimating an ARCH model. This approach was proposed by Engle (1982) to improve the analysis of financial data. It has since proven itself invaluable in modeling any time-series variable in which there are periods of turbulence followed by relative calm at some point. Hourly temperature is one of those variables. Those tempted to claim otherwise are cheerfully invited to consult the book entitled "Environmental Econometrics Using Stata," authored by Baum and Hurn (2021).

The second estimation stage also recognizes that the temperature in hour t is not statistically independent from the temperature outcomes in previous hours, as seen in Figure 9. As suggested in the previous section, this is done using an ARMAX specification. In this case, the transformed explanatory variables from the first stage (e.g.,  $Solar_t^{1/4}$ ) are the exogenous inputs. Given this specification, the disturbance terms are presumed to follow an ARMA specification that models the autocorrelations reported in Figure 9. The ARMA specification applied in this paper is not parsimonious because the autocorrelative process in Figure 9 is not short in duration. It is recognized that this approach runs counter to the traditional time-series philosophy (Box and Jenkins, 1976, p. 17), which suspected that there was more room for prediction errors when more time-series parameters were estimated (Hamilton, 1994, p. 106). The view here is that the goal of predictive accuracy can sometimes be enhanced by including more ARMA terms. This approach makes sense given the long memory property of the autocorrelations evidenced in Figure 9 and the high level of variability in temperature, as evidenced by Figure 5. The heteroskedasticity is modeled as a function of the solar zenith angle, the hour of the day, the day of the year, the year of the sample, and the following variables:  $\sqrt{CO2_{t-1}}, \sqrt{\text{Solar}_t}$ . Instead of assuming that hourly temperature is independent of the conditional variance. the model permits the data to speak for itself on this issue. This linkage is relevant if the level of a variable depends on the variance in the disturbance term. The ARCH-in-mean model introduced by Engel et al. (1987) offers an approach to estimate this linkage.

The possible merits of representing the explanatory variables using a nonlinear specification are addressed using the multivariable fractional polynomial (MFP) methodology (Royston and Sauerbrei, 2008). Its application includes Forbes and St Cyr (2017, 2019) and Forbes and Zampelli(2019, 2020). The methodology considers the effects of nonlinear transformations of the explanatory variables. In the present case, the MFP results suggest the following specification:

 $\begin{aligned} \ln \text{Temp}_{t} &= \alpha_{0}^{'} + \alpha_{1}^{'} \text{ZeroSolar}_{t} + \alpha_{2}^{'} \text{Solar}_{t}^{1/4} + \alpha_{3}^{'} (\text{CO2}_{t-1} * \text{ZeroSolar}_{t})^{3} \\ &+ \alpha_{4}^{'} (\text{CO2}_{t-1} * \text{PosSolar}_{t})^{1/4} + \alpha_{5}^{'} (\text{Solar}_{t} * \text{CO2}_{t-1})^{1/4} + \sum_{h=1}^{9} \beta_{h}^{'} \text{Angle}_{h} \\ &+ \sum_{i=2}^{24} \phi_{i}^{'} \text{HourofDay}_{i} + \sum_{j=2}^{365} \gamma_{i}^{'} \text{DOY}_{j} + \sum_{k=1985}^{2014} \delta_{k}^{'} \text{Year}_{k}(2) \end{aligned}$ 

Please note that  $\alpha'_1$ ,  $\alpha'_2$ , and  $\alpha'_3$  etc. are the estimated coefficients in this specification. Least squares estimation of (2) produces a seemingly respectable level of explanatory power, the R<sup>2</sup> being about 0.831. However, a Portmanteau test for autocorrelation (Box and Pierce, 1970; Ljung and Box, 1978) reveals that the residuals are highly autocorrelated. Consistent with Forbes and St. Cyr (2019, p.17), for lags one through 100, the *P* values are less than 0.0001. The null hypothesis of no ARCH effects is rejected with a *P*- value less than 0.0001. Consistent with these issues, the least-squares model is not useful. This finding is supported by out-of-sample predictions over the period 1 Jan 2016 - 31 Aug 2017 time interval that have a root-mean-squared-error (RMSE) of about 5.67 ° C, a value that is clearly indicative of a suboptimal prediction process.

ARCH/ARMAX methods can generate predictions that are much more accurate than the predictions from a least-squares model when the dependent variable is autoregressive and heteroskedastic in nature. In this case, the ARCH process's modeled lag lengths are lags 1 and 2. Consideration was given to including additional ARCH terms to model the apparent diurnal pattern of the ARCH process (e.g., 24, 48, 72, 96 etc.). Consideration was also given to employing alternative ARCH and GARCH specifications. These approaches were abandoned due to model convergence issues. The modeled lag lengths for the AR process are 1 through 12, 23, 24, 25, 26, 47, 48, 49, 71, 72, 73, 96, 97, 120, 121, 144, 145 167, 168, 169, 192, 193, 216, 240, 264, 288, 312, 335, 336, 337, 360, 384, 408, 432, 456, 480, 600, 671, 672, 673, 840, and 960. The MA modeled lag lengths are 1 through 25, 48, 49, 71, 72, 73, 96, 97, 120, 121, 144, 145, 167, 168, 169, 192, 193, 216, 240, 264, 288, 312, 335, 336, 337, 360, 384, 408, 432, 456, 480, 600, 671, 672, 673, 840, and 960.

Equation (2) was estimated assuming that the residual error terms correspond to the Student t distribution instead of the more typical Gaussian distribution. This approach is believed to be justified by the highly volatile nature of the weather system in the vicinity of Barrow. One shortcoming in its application here is that the "degrees of freedom" parameter is less than the minimum indicated by Harvey (2013, p. 20). Consideration was given to modeling the residual error terms using the generalized error distribution, but this approach was abandoned due to model convergence issues.

Selected estimates are reported in Table 1. It is revealed that  $\alpha'_2$ , the coefficient corresponding to Solart<sup>1/4</sup> is positive and highly statistically significant. The CO<sub>2</sub> coefficients  $\alpha'_3$  and  $\alpha'_4$  are also positive and highly statistically significant while  $\alpha'_5$  is negative and highly statistically significant. These findings are consistent with the view that CO<sub>2</sub> concentrations have implications for hourly temperature but do not address the magnitude. Concerning the possible non-anthropomorphic drivers of temperature, it is interesting to note that 16 of the 30 variables in question are statistically significant. With 2015 being represented in the constant term, negative values for a year are consistent with higher predicted temperatures in 2015 than in the year in question. There are 13 such cases. For these cases, the coefficients' median value is -0.00543, a value that hardly seems important.

The model's explanatory power based on the estimated structural parameters ( all the parameter estimates ) is 0.8105 (0.9968.) Those who believe that the latter level of explanatory power is somehow "too outstanding to be true," are cheerfully invited to reinspect Figure 9 and contemplate the concept of autocorrelation and how modeling this autocorrelation can affect a model's level of explanatory power. In any event, the view here follows Hyndman and Athanasopoulos (2018, 3.4), who note that true adequacy... " can only be determined by considering how well a model performs on new data that were not used when fitting the model." It is also noted that even though a model's  $\mathbb{R}^2$  equivalence is a well-recognized measure of model adequacy, a good case can be made that achieving white noise in the residuals is also important ( Becketti, 2013, p. 256; Kennedy, 2008, p. 315; and Granger and Newbold, 1974, p. 119). To assess whether this measure of adequacy is achieved, Portmanteau tests for autocorrelation were conducted for the hourly lags 1 through 100, 192, 284, and 672. At lag 1, the *P*- value is 0.1958. For the remaining 111 lags that were assessed, the *P*-values are less than .05, thereby rejecting the null hypothesis of a white noise error structure.

Variable	Estimated Coefficient	Absolute Value of the t-Statis
Constant term	-84.5387	3.41
$ m ZeroSolar_t$	0.053421	9.25
$\mathrm{Solar_t}^{1/4}$	0.01102	11.23
$(CO2_{t-1}*ZeroSolar_t)^3$	7.70E-11	7.57
$(\text{CO2}_{t-1}^{*}\text{PosSolar}_{t})^{1/4}$	0.01296	9.04
$(Solar_t * CO2_{t-1})^{1/4}$	-0.00232	10.42
Year <sub>1985</sub>	-0.01111	9.96
Year <sub>1986</sub>	-0.00371	2.36
Year <sub>1987</sub>	-0.00983	6.91
Year <sub>1988</sub>	-0.00808	6.87
Year <sub>1989</sub>	-0.00498	1.76
Year <sub>1990</sub>	-0.0033	1.47

Variable	Estimated Coefficient	Absolute Value of the t-Statist
Year <sub>1991</sub>	-0.00285	1.82
Year <sub>1992</sub>	-0.00664	2.21
Year <sub>1993</sub>	-0.00265	2.52
Year <sub>1994</sub>	-0.00339	2.47
Year <sub>1995</sub>	-0.00384	4.43
Year <sub>1996</sub>	-0.00305	1.73
Year <sub>1997</sub>	0.001996	1.06
Year <sub>1998</sub>	0.005733	3.48
Year <sub>1999</sub>	-0.00766	4.34
Year <sub>2000</sub>	-0.00543	4.26
Year <sub>2001</sub>	-0.00359	2.97
Year <sub>2002</sub>	0.002124	0.61
Year <sub>2003</sub>	-0.00658	3.21
Year <sub>2004</sub>	-0.00449	4.07
Year <sub>2005</sub>	-0.00211	1.11
Year <sub>2006</sub>	0.000883	0.33
Year <sub>2007</sub>	0.005622	4.31
Year <sub>2008</sub>	1.92E-06	0
Year <sub>2009</sub>	0.002597	1.98
Year <sub>2010</sub>	0.000847	0.38
Year <sub>2011</sub>	0.001634	0.23
Year <sub>2012</sub>	-0.00044	0.22
Year <sub>2013</sub>	0.001147	0.46
Year <sub>2014</sub>	0.002601	1.40
Number of Observations	228,085	
R-Square equivalence based on the full model	0.9968	
R-Square equivalence based on the model's structural component.	0.8105	

Regarding the binary variables not reported above, 336 of the 364 day-of-the-year coefficients are statistically significant, wh

### 7 The Model's Out-of-Sample Performance

The out-of-sample evaluation period consists of 13,175 hours over the 1 Jan 2016 to 31 Aug 2017 time interval. Recalling that the dependent variable in the model is the natural logarithm of temperature measured in Kelvin, it might seem that a simple retransformation would yield the optimal predicted value. Unfortunately, merely taking the antilogarithm of the predicted natural logarithm of temperature measured in Kelvin may result in a biased temperature prediction (Granger and Newbold, 1976, pp. 196-197). This bias is easily resolved when the error distribution is Gaussian using a method presented by Guerrero (1993). Given the non-Gaussian nature of the error distribution in this case, the matter was resolved by estimating a post-processing regression without a constant term using all of the observations in the sample. The explanatory variable in this post-processing regression is the hourly temperature measured in Kelvin, while the explanatory variable in this regression is the antilog of the transformed predicted values. The estimated coefficient corresponding to the explanatory variable equals 0.9999895. The associated R-Square equals 1.0000. The estimated parameter from this regression was used to detransform the out-of-sample transformed predicted temperature values.

The out-of-sample predictions were compared with the ERA5 predictions for the same general location. For those unfamiliar with the ERA5 modeling results, it was produced by the Copernicus Climate Change Service at ECMWF. In a significant advance from its earlier databases, it reports hourly values across the globe. The ERA5 hourly temperature values for the Barrow location were obtained from Meteoblue (https://content.meteoblue.com/en/specifications/data-sources/weather-simulation-data/reanalysis-datasets).

The out-of-sample temperature predictions from the ARCH/ARMAX model presented in this paper have a predictive R-square of 0.9962. The predictions are visually more accurate than the ERA5 values for the same general location (Figure 10), although it should be noted that the ERA5 values correspond to a grid that includes land and ocean while Barrow represents a land location within that grid. Nevertheless, the ERA5 values may serve as a useful benchmark for the ARCH/ARMAX out-of-sample predictions. Regarding the RMSEs, the predictions associated with the ARCH/ARMAX model have an RMSE equal to about 0.682 °C, while the ERA5 outcomes have an RMSE of about 3.117°C. Interestingly, an ordinary least-squares estimation of the ERA5 predictions indicates that the prediction errors are not purely random. Specifically, the prediction error is conditional on the magnitude of the predicted temperature and lagged value of the CO<sub>2</sub> concentration. The latter finding is consistent with the central thesis of this paper. Following Granger's discussion of prediction errors (1986, p. 91), both of these findings suggest a pathway to improving the accuracy of the ERA5 predictions.

The out-of-sample temperature predictions from the ARCH/ARMAX model are significantly degraded when the estimated effects of CO<sub>2</sub> are ignored (Figure 11). The differential in predictive accuracy is visually apparent if one inspects the vertical distance between the scatter points and the 45° line representing the relationship between predicted and actual temperature when the predictions are perfect. As reported above, the full model presented in this paper has an RMSE equal to 0.682 °C over the evaluation period, constraining the CO<sub>2</sub> estimated effects to be equal to zero results in predictions with an RMSE equal to 3.379 °C.

The out-of-sample analysis is supportive of the earlier discussion indicating the unimportance of factors other than  $CO_2$  and the total downward solar irradiance being drivers of the increase in annual temperature over the sample period. Specifically, using the full model, the mean predicted temperature over the evaluation period equals - 8.725218 °C. The mean predicted temperature over the evaluation period is -8.725221 °C if the estimated effects of the binary variables for 1986 through 2014 are constrained to equal zero. In short, the binary variables that control for the possibility of annual temperature being affected by factors other than  $CO_2$  or total downward solar irradiance have virtually no effect on the out-of-sample predicted temperature. Interestingly, the mean actual temperature over the evaluation period equals -8.712713°C, a very close value to the mean of the predicted values.



Figure 10. The ERA5 and the ARCH/ARMAX prediction errors, 1 Jan 2016 – 31 Aug 2017.



Figure 11. The ARCH/ARMAX model predictions with and without the  $CO_2$  estimated effects and the actual temperature outcomes, 1 Jan 2016 – 31 Aug 2017.

The structural predictions are less accurate than the predictions from the full model but may yield useful insights. The predictions from the structural model have an RMSE equal to 5.21 °C while constraining the  $CO_2$  estimated effects to be equal to zero results in predictions with an RMSE equal to 8.29°C (Figure

12). In short, constraining the estimated effects of  $CO_2$  to be equal to zero reduces the structural model's predictive accuracy. In terms of temperature, the predicted level is significantly lower when the estimated structural effects of  $CO_2$  are ignored (Figure 13). Observe that the difference in the mean levels of predicted temperature is nontrivial.



Figure 12. The RMSEs in the out-of-sample structural predictions



Figure 13. The out-of-sample structural predictions of temperature (°C)

#### 8 Summary and Conclusion

This paper employed an ARCH/ARMAX model with statistical controls for total downward solar irradiance and 426 binary variables to examine the relationship between CO<sub>2</sub> concentrations and hourly temperature at the Barrow Atmospheric Observatory in Alaska. The model was estimated using hourly data over the time interval of 1 Jan 1985 - 31 Dec 2015. The model was evaluated using hourly data from 1 Jan 2016 through 31 Aug 2017. The predictive R-square equivalence of 0.9962 over the evaluation period suggests that the model has reduced the attribution challenge associated with the significant natural meteorological variability in the Arctic. Consistent with this view, the predictions over the evaluation period are more accurate than the highly regarded ERA5 values for the same general vicinity. Thus, though the model fails to achieve the metric of "white noise" in the standardized residuals, the accuracy of its predictions over the evaluation period indicates that the model is "useful." These results are consistent with the physics that indicates that rising  $CO_2$  concentrations have consequences for temperature, a point that even climate deniers such as Richard Lindzen, William Happer, Roy Spencer, Patrick Michaels, and the other members of the  $CO_2Coalition$  have conceded. What is different is that the model also offers useful insights into the magnitude of the relationship between CO<sub>2</sub> concentrations and hourly temperature. Specifically, the predictions over the evaluation period are significantly more accurate when they reflect the estimated and statistically significant  $CO_2$  coefficients compared to when those coefficients are ignored. The out-of-sample results indicate that CO<sub>2</sub>concentrations have nontrivial implications for hourly temperature. The modeling results also addressed the possible contribution of factors other than  $CO_2$  being drivers of increased temperature over the sample. The mean of the out-of-sample predicted temperature over the evaluation period is not materially affected by these variables, even though some of those variables are statistically significant.

Given that all models are "wrong," it is a picayune task to dismiss the estimation results reported in Table 1. It is much more challenging to rationally dismiss the implications of the large decline in the out-of-sample predictive accuracy when the estimated CO<sub>2</sub>effects are ignored. One possibility is that some unknown natural factor at work is the true culprit of the decline in predictive accuracy. While climate deniers may find this an attractive explanation for the results presented in this paper, the model's high level of predictive out-of-sample accuracy suggests that unknown factors are not an important driver of temperature. There is also the point that attributing the large decline in the out-of-sample predictive accuracy when the estimated CO<sub>2</sub> effects are ignored to an "unknown variable" is highly likely to represent obscurantism as opposed to a conclusion that represents the best of all competing explanations as explained by Lipton (2004, p. 56). In short, the beliefs of the climate change deniers are not supported by the hourly temperature data at NOAA's Barrow Observatory in Alaska. Considering the inadequate results of COP26, this suggests that the current outlook for the Earth's future is quite grim. Research that further illuminates the shortcomings of the views by climate deniers might help matters. One approach being considered is an analysis of the drivers of the hourly surface energy imbalance, a metric that is easily understood as being important but that climate deniers almost never mention. This research path appears feasible using the methods presented here in light of a preliminary analysis indicating that the hourly surface energy imbalance at Barrow and other locations is autoregressive and heteroskedastic. It is not overly optimistic to believe that modeling these properties will facilitate the recognition of  $CO_2$ 's "signal" in the data.

#### Acknowledgments

The paper's findings are based on the data collected at the Barrow Atmospheric Observatory. I thank the Earth System Research Laboratory (ESRL), Global Monitoring Division (GMD), of the National Oceanic and Atmospheric Administration (NOAA) for supporting the operations of this observatory. I am also grateful to NOAA's National Weather Service and National Centers for Environmental Information for their stewardship of the PABR temperature data. The unit root conclusions rely partly on the hourly temperature data reported by the Syowa observatory in Antarctica. I am thankful to the National Institute of Polar Research in Japan for supporting the operations of this observatory. Earlier versions of this paper were presented at the ESIPP/UCD weekly seminar and the ESRI/UCD policy conference. I

thank the participants for their feedback. The research was also presented as an e-poster at the 2020 AGU Fall meetings in a session entitled, "Convergence Research in Climate Science: How to Move Beyond Disciplinary Silos." I thank the participants of that session for their comments. I thank Jeffery Wooldridge, Nick Cox, and Maarten Buis for sharing their econometric insights. I thank Rick Thoman for sharing his meteorological insights. I also thank Chris St. Cyr and Dina Tady for their candid comments on an earlier draft. Any errors are the full responsibility of the author.

#### ORCID

Kevin F. Forbes, https://orcid.org/0000-0002-9521-6845

#### **Conflict of Interest**

The author declares no conflicts of interest relevant to this study.

#### **Data Availability Statement**

Data used in this research and reproducing STATA codes are deposited on Zenodo at 10.5281/zenodo.5833580.

#### References

American Association for the Advancement of Science, (2009). 1021Climate Letter,

https://www.aaas.org/sites/default/files/1021climate\_letter.pdf

American Association for the Advancement of Science, (2016). Thirty-One Top Scientific Societies Speak with One Voice on Global Climate Change. Available at https://www.aaas.org/news/thirty-one-top-scientific-societies-speak-one-voice-global-climate-change

AMAP, 2019. AMAP Climate Change Update 2019: An Update to Key Findings of Snow, Water, Ice and Permafrost in the Arctic (SWIPA) 2017. Arctic Monitoring and Assessment Programme (AMAP), Oslo, Norway. 12 pp.

Australian Academy of Sciences et al. (2001) The Science of Climate Change, *Science* 18 May 2001: Vol. 292, Issue 5520, pp. 1261 DOI: 10.1126/science.292.5520.1261 https://science.sciencemag.org/content/292/5520/1261

Baum, C. F., and Hurn, S. (2021), *Environmental Econometrics Using Stata*, Stata Press, College Station, Texas

Becketti, S. (2013). Introduction to time series using stata. College Station, TX: Stata Press.

Box, G.E.P. and Jenkins, G.M. (1976) *Time Series Analysis: Forecasting and Control*, rev. ed., San Francisco: Holden Day

Box, G. E. P., and D. A. Pierce (1970). Distribution of residual autocorrelations in autoregressive-integrated moving average time series models. *Journal of the American Statistical Association* 65: 1509–1526.

Box, G. E. P., Hunter, J. S. & Hunter, W. G. (2005), *Statistics for Experimenters* (2nd ed.), John Wiley & Sons.

# Box, J. E. et al. (2019). Key indicators of Arctic climate change: 1971–2017

Environ. Res. Lett .14 045010 https://iopscience.iop.org/article/10.1088/1748-9326/aafc1b

Brooks, C., (2019). Introductory Econometrics for Finance ,  $4^{\rm th}$  edition, New York: Cambridge University Press

CO<sub>2</sub> Coalition. (2015). Carbon Dioxide Benefits the World https://co2coalition.org/publications/carbon-dioxide-benefits-the-world-see-for-yourself/

Elliot, G., Rothenberg, T.J. and Stock, J.H. (1996). "Efficient Tests for an Autoregressive Unit Root," *Econometrica*, 64, 813-836

Engle, R. F. (1982). Autoregressive conditional heteroscedasticity with estimates of the variance of United Kingdom inflation. *Econometrica* 50: 987–1007.

Engle, R.F., Lilien, D.M. & Robins, R.P. (1987) Estimating Time Varying Risk Premia in the Term Structure: The Arch-M Model. *Econometrica*, 55, 391-407. https://doi.org/10.2307/1913242

EPIC (2018). New Poll: Nearly half of Americans are more convinced than they

were five years ago that climate change is happening, with extreme weather driving their views. Available at: https://apnorc.org/projects/is-the-public-willing-to-pay-to-help-fix-climate-change/

EPA (2021). Climate Change Indicators in the United States. Explanation of the data available at https://www.epa.gov/sites/production/files/2021-04/documents/permafrost\_td.pdf

Raw data available at: https://www.epa.gov/sites/production/files/2021-04/permafrost\_fig-1.csv

Forbes, K. F., & St. Cyr, O. C. (2019). The Challenge Posed by Space Weather to High-Voltage Electricity Flows: Evidence From Ontario, Canada, and New York State, USA. *Space Weather*, 17. https://doi.org/10.1029/2019SW002231

Forbes, K. F., & St. Cyr, O. C. (2017). The challenge posed by geomagnetic activity to electric power reliability: Evidence from England and Wales. *Space Weather*, 15 (10), 15–1430. https://doi.org/10.1002/2017SW001668

Forbes, K.F., Zampelli, E.M. (2020). Accuracy of wind energy forecasts in Great Britain and prospects for improvement, *Utilities Policy*, Volume 67 Page: 101111

Forbes, K.F., Zampelli, E.M., (2019). Wind energy, the price of carbon allowances, and CO2 emissions: evidence from Ireland. *Energy Pol*. 133.

Granger, Clive W.J., (1986). *Forecasting in Business and Economics*, second ed. Economic Theory, Econometrics, and Mathematical Economics.

Granger, C. W. J. & Newbold, P. (1976) Forecasting transformed series, *Journal of the Royal Statistical Society*, B-38, 189-203.

Granger, C. W. J., & Newbold, P. (1974). Spurious regressions in econometrics. *Journal of Econometrics*, 2(2), 111–120. https://doi.org/10.1016/0304-4076(74)90034-7

Guerrero, Victor M. (1993). "Time-series analysis supported by power transformations." *Journal of Fore-casting* 12(1): 37–48. http://dx.doi.org/10.1002/for.3980120104.

Hamilton, J. D. (1994). Time Series Analysis, Princeton, NJ: Princeton University Press

Harvey, A. C. (2013). Dynamic models for volatility and heavy tails: With applications to financial and economic time series. New York: Cambridge University Press. https://doi.org/10.1017/CBO9781139540933

Herbert, G.A., Green, E.R., Harris, J.M., Koenig, G.L., Roughton, S.J., and Thaut, K.W., (1986). Control and monitoring instrumentation for the continuous measurement of atmospheric CO2 and meteorological variables, J. Atmos. Oceanic Technol., 3, 414-421.

DOI:https://doi.org/10.1175/1520-0426(1986)003<0414:CAMIFT>2.0.CO;2

Hyndman, R.J., Athanasopoulos, G., 2018. Forecasting: Principles and Practice, second ed. OTexts, Melbourne, Australia. OTexts.com/fpp2.

IPCC, (2013). Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker, T.F., Qin, D., Plattner, G.-K., Tignor, M., Allen, S.K., Boschung, J., Nauels, A., Xia, Y., Bex V., & Midgley P.M., (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

Kennedy, P. (2008). A Guide to Econometrics, sixth edition, Malden Massachusetts: Blackwell Publishing

Leahy, P., (2019). Irish Times poll: Climate change' most serious issue' for majority of voters, Irish Times https://www.irishtimes.com/news/politics/irish-times-poll-climate-change-most-serious-issue-for-majority-of-voters-1.4051713

Lindzen, R., (2020). On Climate Sensitivity, Available at: https://co2coalition.org/publications/on-climate-sensitivity/

Lipton, P., 2004. Inference to the Best Explanation . London Routledge.

Ljung, G.M., Box, G.E.P., 1978. On a measure of lack of fit in time series models. Biometrika 65, 297–303

Markon, C., Gray, S., Berman, M., Eerkes-Medrano, L., Hennessy, T., Huntington, H., Littell, J., Mc-Cammon, M., Thoman, R., & Trainor, S. (2018) Alaska. In Impacts, Risks, and Adaptation in the United States: Fourth National Climate Assessment, Volume II [Reidmiller, D.R., Avery, C.W., Easterling, D.R., Kunkel, K.E., Lewis, K.L.M., May-cock, T.K. & Stewart B.C. (eds.)]. U.S. Global Change Research Program, Washington, DC, USA, pp. 1185–1241. doi: 10.7930/NCA4.2018.CH26 Available at: https://nca2018.globalchange.gov/chapter/alaska

National Academies of Science (2005). Joint science academies' statement: Global response to climate change https://sites.nationalacademies.org/cs/groups/internationalsite/documents/webpage/international\_-080877.pdf

Peterson, J.T., Komhyr, W.D., Waterman, L.S., Gammon, R.H., Thoning, K.W., and Conway, T.J. (1986). Atmospheric CO2 variations at Barrow, Alaska, 1973-1982, *J. Atmos. Chem.*, 4, 491-510. DOI:10.1007/BF00053848

Pithan, F., & Mauritsen, T. (2014). Arctic amplification dominated by temperature feedbacks in contemporary climate models. *Nature Geoscience*, 7 (3), 181–184. https://doi.org/10.1038/ngeo2071

Post, E., Alley, R. B., Christensen, T. R., Macias-Fauria, M., Forbes, B. C., Gooseff, M. N.,

Iler, A., Kerby, J. T. , Laidre, K. L., Mann, M. E., Olofsson, J., Stroeve, J. C., Ulmer, F., Virginia, R. A., & Wang, M. (2019). The polar regions in a 2°C warmer world. *Sci. Adv.* **5** , eaaw9883*DOI:* 10.1126/sci-adv.aaw9883

STATA, 2021. Autoregressive conditional heteroskedasticity (ARCH) family of estimators, in Time-Series Reference Manual 17 https://www.stata.com/manuals/ts.pdf

Romanovsky, V.E., Isaksen, K., Drozdov, D.S., Anisimov, O., Instanes, A., Leibman, M., Mcguire A. D., Shiklomanov, N., Smith, S., & Walker D. (2017) Changing permafrost and its impacts. In: Snow, Water, Ice and Permafrost in the Arctic (SWIPA) 2017. pp. 65-102. Arctic Monitoring and Assessment Programme (AMAP), Oslo, Norway. https://www.amap.no/documents/download/2987/inline

Royston, P., & Sauerbrei, W. (2008). Wiley series in probability and statistics. In *Multivariable model*building: A pragmatic approach to regression analysis based on fractional polynomials for modelling continuous variables. Chichester, UK: John Wiley. https://doi.org/10.1002/9780470770771.scard

Schuur, E. A. G., Bracho, R., Celis, G., Belshe, E. F., Ebert, C., Ledman, J., et?al. (2021). Tundra underlain by thawing permafrost persistently emits carbon to the atmosphere over

15 years of measurements. Journal of Geophysical Research: Biogeosciences , 126, e2020JG006044. https://doi.org/10.1029/2020JG006044 Tans, P. and Thoning K,(2020). How we measure background CO2 levels on Mauna Loa. NOAA ESRL Global Monitoring Division. Available at:

https://gml.noaa.gov/ccgg/about/co2\_measurements.html

Taylor, P.C., Maslowski, W., Perlwitz, J., & Wuebbles, D. J., (2017). Arctic changes and their effects on Alaska and the rest of the United States. In: Climate Science Special Report: Fourth National Climate Assessment, Volume I [Wuebbles, D.J., Fahey, D.W., Hibbard, K.A., Dokken, D.J., Stewart, B.C. & Maycock, T.K. (eds.)]. U.S. Global Change Research Program, Washington, DC, USA, pp. 303-332, doi: 10.7930/J00863GK

Thoman, R. L., J. Richter-Menge, and M. L. Druckenmiller, Eds., (2020). Arctic Report Card (2020), https://doi.org/10.25923/mn5p-t549

Thoning, K.W., Crotwell, A.M., and Mund J.W. (2021), Atmospheric Carbon Dioxide Dry Air Mole Fractions from continuous measurements at Mauna Loa, Hawaii, Barrow, Alaska,

American Samoa and South Pole. 1973-2020, Version 2021-08-09, National Oceanic and Atmospheric Administration (NOAA), Global Monitoring Laboratory (GML), Boulder, Colorado, USAhttps://doi.org/10.15138/yaf1-bk21

FTP path:ftp://aftp.cmdl.noaa.gov/data/greenhouse\_gases/co2/in-situ/surface

Trout, M. D. (2006). Regression, 10k Rule of Thumb for, *Encyclopedia of Statistical Sciences*, John Wiley & Sons, Inc.

UNFCCC Secretariat, (2021a). Nationally determined contributions under the Paris Agreement: Synthesis Report by the Secretariat

https://unfccc.int/sites/default/files/resource/cma2021\_08E.pdf

UNFCCC Secretariat, (2021b). 4 Key Achievements of COP26, https://unfccc.int/news/4-key-achievements-of-cop26

U.S. Department of Commerce, National Oceanic and Atmospheric Administration National Centers for Environmental Information, (2021) Barrow Airport Global Summary of the Year 1951 - 2020. Available at https://www.ncdc.noaa.gov/cdo-web/search?datasetid=GSOY

U.S. Department of Commerce (1951) Local Climatological Data: Annual Summary with Comparative Data, Barrow Alaska

Available at: https://www.ncdc.noaa.gov/IPS/lcd/lcd.html?\_finish=0.9630710752323048

Vasel, B., Schultz, C., Schnell, R., Stanitski, D., and Thomas, B., (2020). New Arctic Research Facility Opens Door to Science Collaborations. Arctic Report Card: 2020 Update. NOAA Arctic Program. DOI: 10.25923/24rn-c757. Accessed at: https://arctic.noaa.gov/Report-Card/Report-Card-2020/ArtMID/7975/ArticleID/908/New-Arctic-Research-Facility-Opens-Door-to-Science-Collaborations

Winton, M. (2006). Amplified Arctic climate change: What does surface albedo feedback have to do with it? *Geophysical Research Letters*, 33 (3). https://doi.org/10.1029/2005gl025244

Yang, J., Gounaridis, D., Liu, M., Bi, J., & Newell, J. P. (2021). Perceptions of

climate change in China: Evidence from surveys of residents in six cities. Earth's

Future, 9, e2021EF002144. https://doi.org/10.1029/2021EF002144

Wilks, D. (2019), Statistical Methods in the Atmospheric Sciences, Elsevier, Cambridge, MA.

1 2 3 4	Hourly temperature data do not support the views of the Climate Deniers: Evidence from Barrow Alaska
5	
6	Kevin F. Forbes, Ph.D <sup>1</sup>
7	
8	<sup>1</sup> Energy and Environmental Data Science, Malahide, Ireland
9	
10	Corresponding author: Kevin F. Forbes ( <u>Kevin.F.Forbes@EEDS.Solutions)</u>
11	Key Points:
12	
13	1) At NOAA's Barrow Observatory in Alaska, the annual temperature during 2015-2020
14	was about 3.37 °C higher than in 1985-1990.
15	
16	2) Virtually all the upward trend in annual temperature through 2015 can be attributed to
17	higher CO <sub>2</sub> concentrations.
18	
19 20	3) The model's out-of-sample predictions are more accurate if the estimated associations between $CO_2$ and temperature are not ignored.

# Abstract

22 23 Survey evidence has indicated that a significant percentage of the population does not fully embrace the scientific consensus regarding climate change. This paper assesses whether the 24 25 hourly temperature data support this denial. The analysis examines the relationship between hourly CO<sub>2</sub> concentration levels and temperature using hourly data from the NOAA-operated 26 Barrow observatory in Alaska. At this observatory, the average annual temperature over the 27 28 2015-2020 period was about 3.37 °C higher than in 1985–1990. A time-series model to explain hourly temperature is formulated using the following explanatory variables: the hourly level of 29 total downward solar irradiance, the CO<sub>2</sub> value lagged by one hour, proxies for the diurnal 30 variation in temperature, proxies for the seasonal temperature variation, and proxies for possible 31 non-anthropomorphic drivers of temperature. The purpose of the time-series approach is to 32 capture the data's heteroskedastic and autoregressive nature, which would otherwise "mask" 33 CO<sub>2</sub>'s "signal" in the data. The model is estimated using hourly data from 1985 through 2015. 34 The results are consistent with the hypothesis that increases in CO<sub>2</sub> concentration levels have 35 nontrivial consequences for hourly temperature. The estimated annual contributions of factors 36 exclusive of CO<sub>2</sub> and downward total solar irradiance are very small. The model was evaluated 37 38 using out-of-sample hourly data from 1 Jan 2016 through 31 Aug 2017. The model's out-ofsample hourly temperature predictions are highly accurate, but this accuracy is significantly 39 degraded if the estimated CO<sub>2</sub> effects are ignored. In short, the results are consistent with the 40 41 scientific consensus on climate change.

42

21

43 44

45

# 46 Plain Language Summary

According to the IPCC and other scientific organizations, "it is extremely likely that human 47 influence has been the dominant cause of the observed increase in global temperatures since the 48 mid-20th century." However, a significant percentage of the population does not fully embrace 49 this consensus. Using data from the Barrow Atmospheric Observatory, this paper assesses 50 whether the hourly temperature data support this apparent denial. It is first noted that the 51 average annual temperature at Barrow over the 2015-2020 period was about 3.37 °C higher than 52 in the 1985-1990 period. The formal analysis employs hourly solar irradiance, CO<sub>2</sub>, and 53 temperature data. The model controls for possible non-anthropomorphic drivers of annual 54 temperature and other factors. The model was estimated using hourly data over the time interval 55 1 Jan 1985 through 31 Dec 2015. The estimated annual effects of CO<sub>2</sub> are significant in 56 magnitude, while the non-anthropomorphic drivers exclusive of solar irradiance are quantitively 57 unimportant. The model is evaluated over the 1 Jan 2016 through 31 Aug 2017 time interval. 58 The model's out-of-sample hourly temperature predictions are highly accurate, but this accuracy 59 is degraded if the estimated CO<sub>2</sub> effects are ignored. In short, the results are consistent with the 60 scientific consensus on climate change. 61

63	
64	Index Terms
65	6620 Science Policy
66	1630 Impacts of Global Change
67	1616 Climate Variability
68	9315 Arctic Region
69 50	32/0 Time series analysis
70	1986 Statistical methods: Inferential
/1	
72 73	Key Words:
74	CO <sub>2</sub> Concentrations, Hourly Temperature, Downward total solar irradiance, Climate Change,
75	Arctic Region, Alaska
76	
77	Acronyms: AMAP, Arctic Monitoring and Assessment Program, ARCH, Autoregressive
78	conditional heteroskedasticity; ARMA, autoregressive-moving-average; ARMAX,
79	autoregressive-moving-average with exogenous inputs; ECMWF, European Centre for Medium-
80	Range Weather Forecasts. MFP, multivariable fractional polynomial; RMSE, root-mean-
81	squared-error.
82	
02	
83	1. Introduction
83 84	1. Introduction
83 84 85	<ol> <li>Introduction</li> <li>According to the IPCC, "It is extremely likely that human influence has been the dominant cause</li> </ol>
82 83 84 85 86	<ol> <li>Introduction</li> <li>According to the IPCC, "It is extremely likely that human influence has been the dominant cause of the observed increase in global temperatures since the mid-20th century "(IPCC, 2013, p. 17).</li> </ol>
82 83 84 85 86 87	<ul> <li>1. Introduction</li> <li>According to the IPCC, "It is extremely likely that human influence has been the dominant cause of the observed increase in global temperatures since the mid-20th century "(IPCC, 2013, p. 17). As early as 2001, the science academies of Australia, Belgium, Brazil, Canada, the Caribbean,</li> </ul>
82 83 84 85 86 87 88	<ul> <li>1. Introduction</li> <li>According to the IPCC, "It is extremely likely that human influence has been the dominant cause of the observed increase in global temperatures since the mid-20th century "(IPCC, 2013, p. 17).</li> <li>As early as 2001, the science academies of Australia, Belgium, Brazil, Canada, the Caribbean, China, France, Germany, India, Indonesia, Ireland, Italy, Malaysia, New Zealand, Sweden, </li> </ul>
82 83 84 85 86 87 88 88 89	<ul> <li>1. Introduction</li> <li>According to the IPCC, "It is extremely likely that human influence has been the dominant cause of the observed increase in global temperatures since the mid-20th century "(IPCC, 2013, p. 17). As early as 2001, the science academies of Australia, Belgium, Brazil, Canada, the Caribbean, China, France, Germany, India, Indonesia, Ireland, Italy, Malaysia, New Zealand, Sweden, Turkey, and the United Kingdom all endorsed the IPCC's Third Assessment (<u>Australian Academy</u>)</li> </ul>
<ul> <li>82</li> <li>83</li> <li>84</li> <li>85</li> <li>86</li> <li>87</li> <li>88</li> <li>89</li> <li>90</li> </ul>	1. Introduction According to the IPCC, "It is extremely likely that human influence has been the dominant cause of the observed increase in global temperatures since the mid-20th century "(IPCC, 2013, p. 17). As early as 2001, the science academies of Australia, Belgium, Brazil, Canada, the Caribbean, China, France, Germany, India, Indonesia, Ireland, Italy, Malaysia, New Zealand, Sweden, Turkey, and the United Kingdom all endorsed the IPCC's Third Assessment ( <u>Australian Academy</u> of Sciences et al., 2001). A more recent list of scientific academies that have accepted this view
<ul> <li>82</li> <li>83</li> <li>84</li> <li>85</li> <li>86</li> <li>87</li> <li>88</li> <li>89</li> <li>90</li> <li>91</li> </ul>	<ul> <li>1. Introduction</li> <li>According to the IPCC, "It is extremely likely that human influence has been the dominant cause of the observed increase in global temperatures since the mid-20th century "(IPCC, 2013, p. 17). As early as 2001, the science academies of Australia, Belgium, Brazil, Canada, the Caribbean, China, France, Germany, India, Indonesia, Ireland, Italy, Malaysia, New Zealand, Sweden, Turkey, and the United Kingdom all endorsed the IPCC's Third Assessment (Australian Academy of Sciences et al., 2001). A more recent list of scientific academies that have accepted this view includes the science academies in Japan, Russia, and the USA. (National Academies of Science, Interval)</li> </ul>
<ul> <li>82</li> <li>83</li> <li>84</li> <li>85</li> <li>86</li> <li>87</li> <li>88</li> <li>89</li> <li>90</li> <li>91</li> <li>92</li> </ul>	<ul> <li>1. Introduction</li> <li>According to the IPCC, "It is extremely likely that human influence has been the dominant cause of the observed increase in global temperatures since the mid-20th century "(IPCC, 2013, p. 17).</li> <li>As early as 2001, the science academies of Australia, Belgium, Brazil, Canada, the Caribbean,</li> <li>China, France, Germany, India, Indonesia, Ireland, Italy, Malaysia, New Zealand, Sweden,</li> <li>Turkey, and the United Kingdom all endorsed the IPCC's Third Assessment (<u>Australian Academy</u></li> <li>of Sciences et al., 2001). A more recent list of scientific academies that have accepted this view</li> <li>includes the science academies in Japan, Russia, and the USA. (National Academies of Science,</li> <li>2005). These institutes are not indicating that human activity is only partly responsible for climate</li> </ul>

95	In the United States, a country in which a nontrivial number of climate deniers hold powerful
96	elected positions, a group of 18 highly respected scientific organizations explicitly endorsed the
97	scientific consensus on climate change in a 2009 letter to U.S. policymakers (American
98	Association for the Advancement of Science, 2009). This Letter was released again in 2016 by a
99	larger group of 31 scientific organizations (American Association for the Advancement of Science,
100	2016). The updated Letter makes the following point:
101 102 103 104 105	"Observations throughout the world make it clear that climate change is occurring, and rigorous scientific research concludes that the greenhouse gases emitted by human activities are the primary driver. This conclusion is based on multiple independent lines of evidence and the vast body of peer-reviewed science."
100	AAAS, 2016
108 109	This paper's starting point is the observation that the survey data does not fully reflect the scientific
110	consensus. This paper applies methods developed to address issues in economics and finance to
111	assess whether the temperature data at the Barrow Atmospheric Observatory in northern Alaska
112	supports this view. While some might sharply question the approach employed in this paper
113	because the methodology is "unorthodox" relative to the conventional meteorological framework,
114	it may be worth noting that the methodology applied in this paper has revolutionized the analysis
115	in other sectors when the data are found to be autoregressive and heteroskedastic in nature. One
116	modest example of this is Forbes and Zampelli (2019), who analyzed CO <sub>2</sub> emissions from the Irish
117	
117	power grid using the methods presented in this paper after observing that the emission levels had
117	power grid using the methods presented in this paper after observing that the emission levels had autoregressive and heteroskedastic properties. These properties will be shown to be highly
117 118 119	power grid using the methods presented in this paper after observing that the emission levels had autoregressive and heteroskedastic properties. These properties will be shown to be highly relevant when modeling hourly temperature. Ignoring these properties makes extracting $CO_2$ 's

In terms of organization, section 2 of the paper discusses the survey data. Section 3 121 summarizes the views of individuals identified as being climate deniers within the scientific 122 community. Section 4 discusses the data used in the analysis. To provide context, the trends in 123 hourly temperature, downward total solar irradiance, and CO<sub>2</sub> concentrations at the Barrow 124 Atmospheric Observatory are reported. In response to an assertion about a lack of recent warming 125 126 relative to the pre-1940 period by Lindzen (2020, pp. 12-13), the annual temperature at the nearby Barrow Airport from 1921 through 2020 is reported. The time-series nature of hourly temperature 127 at Barrow is also discussed to facilitate the modeling discussion in the remaining sections of the 128 paper. Section 5 introduces a modeling framework to examine the possible association between 129 CO<sub>2</sub> concentrations and hourly temperature. Section 6 discusses the estimation process and also 130 presents the results. Section 7 evaluates the model. The paper's findings are discussed in section 131 8. 132

- 133
- 134 135

## 2. The Survey Evidence

A 2019 YouGov survey of 30,000 individuals that are believed to be representative of the online 136 population in 28 countries indicated that there were only 14 countries in which 50 % or more of 137 the respondents would agree with the statement that "The climate is changing and human activity 138 is mainly responsible" (Figure 1). A significant number of the respondents indicated that human 139 140 activity is only partly responsible for climate change. For example, while 40% of the respondents in Denmark agreed with the scientific consensus, 48% agreed with the view that "...human activity 141 is partly responsible, together with other factors (emphasis added). In the United Kingdom, 142 51% endorsed the scientific consensus, while 37% believe that human activity is only partly 143 responsible. In China, 45% endorsed the scientific consensus, while 48% believe human activity 144 is only partly responsible. In the USA, 38% endorsed the scientific consensus, 37% reported that 145

- they believe that human activity is only partly responsible for climate change, 9% believe that
- human activity is not a driver of climate change, and 6% reported that they do not believe that the
- climate is changing.

India	71	23	
Thailand	69	27	
Spain	69	27	
Indonesia	69	24	
Italy	66	29	
Vietnam	64	32	
Philippines	62	31	
Singapore	54	39	
Tawain	53	42	
Qater	52	41	
Kuwait	52	34	4
UAE	52	33	6
Great Britain	51	37	
Hong Kong	50	45	
Finland	49	38	5
Germany	49	36	5
France	48	37	4
Malaysia	48	43	
Bahrain	46	41	4
China	45	48	
Australia	44	43	5
Oman	43	47	
Egypt	42	38	7
Denmark	40	48	4
USA	38	37 9	6
Sweden	36	48	6
Saudi Arabia	35	36 7	5
Norway	35	48	8

Source: 10.5281/zenodo.5833580

Figure 1. Responses to a 2019 YouGov survey question posed to 30,000 people in 28 countries. Thinking about the global environment...In general, which of the following statements, if any, best describes your view?" 

While it is tempting to attribute the findings for China in Figure 1 as evidence of a form of climate 159 denial by a large proportion of its population, the recent findings by Yang et al. (2021) would seem 160 to suggest that a sincere misunderstanding of the nature of climate change might be a more 161 important consideration. In other countries, other survey data are largely consistent with the data 162 presented in Figure 1. For example, in a 2019 Irish Times/Iposos MRBI poll (Leahy, P., 2019), 163 164 respondents were asked if they agreed with the following statement: "I don't think climate change will be as bad as some say so I'm not that worried about it." While 57% of the respondents 165 implicitly endorsed the scientific consensus by disagreeing with the statement, 33% agreed. In 166 this same poll, only 44% of the respondents agreed with the statement, "I am okay with the price 167 168 of oil, gas, petrol and diesel increasing to help tackle climate change." This is obviously not a majority and thus represents a challenge to implementing policies to reduce emissions. 169

170

171 A November 2018 survey of 1,202 adults by the Energy Policy Institute at the University of

172 Chicago and the AP-NORC Center yields useful insights (EPIC, 2018). According to this

survey, 57% of the respondents were willing to pay a \$1 monthly fee to combat climate change.

About 23% were willing to pay 40 USD per month. However, 43 percent were unwilling to pay

anything, highlighting the challenge of doing anything significant to reduce emissions.

Acceptance of the view that human activity contributes to climate change was a useful indicatorof whether respondents were willing to pay to reduce emissions.

178

Suggestive of the possible political implications of the polling data, the UNFCCC secretariat
(United Nations Framework Convention on Climate Change) issued a report in September 2021
that indicated that the combined updated Paris Accord pledges fall short of what it will take to
meet the goals of the Paris Accords. Specifically, even with the updated pledges, projected GHG

emissions in 2030 are only about 0.5% lower than in 2010, which is far lower than what it would 183 take to limit global warming to below two °C (UNFCCC Secretariat, 2021a). The COP26 184 185 meetings that were held in November of 2021 have done little to improve the prospects that the goals of the 2015 Paris Accords will be met. The United States did announce its good intentions, 186 but climate deniers will most likely make those goals very difficult to achieve. The conference 187 faced other challenges including objections to phasing out coal. While the conference made 188 progress in the areas of carbon markets and finance, the fact remains that there is a significant 189 emissions gap (UNFCCC Secretariat, 2021b). 190

#### **3** The Views of the Climate Deniers from within the Scientific Community 191

Somewhat surprisingly, some prominent individuals from within the scientific community who 192 193 have been labeled as climate deniers have actually conceded that increases in CO2 concentrations have consequences for surface warming. For example, the  $CO_2$  Coalition (2015), a sharp critic of 194 the scientific consensus, whose members include the well-known influencers Richard Lindzen, 195 Patrick Michaels, Roy Spence, and William Happer, has explicitly acknowledged the greenhouse 196 effect. It notes that predicting greenhouse-induced warming is difficult because atmospheric 197 processes are very complicated. It then pivots back and reports that it believes that the data 198 suggests that the warming associated with a doubling of  $CO_2$  levels will be very modest. In its 199 words, 200

201

"Basic physics implies that more atmospheric CO<sub>2</sub> will increase greenhouse warming. However, atmospheric processes are so complicated that the amount of 202 warming cannot be reliably predicted from first principles. Recent observations of 203 the atmosphere and oceans, together with geological history, point to very modest 204 warming, about 1 C (1.8 F) if atmospheric CO<sub>2</sub> levels are doubled." 205 206  $CO_2$  Coalition, 2015 207 208

The  $CO_2$  Coalition's assertion that the warming associated with a doubling of  $CO_2$  will be modest appears to be largely premised on a belief that the recent warming is about the same as before the 1940s (Lindzen, 2020, pp. 12-13). As will be seen, this belief is not supported by the data in northern Alaska.

215

# **4 An Overview of the Changing Climate in Northern Alaska**

The study employs temperature, solar radiation, and CO<sub>2</sub> data reported by the Barrow (BRW) 216 Atmospheric Observatory. This is one of the baseline observatories of the Earth System Research 217 Laboratory (ESRL), Global Monitoring Division (GMD), of the National Oceanic and 218 Atmospheric Administration (NOAA). It is located near sea level about 8 km east of Utgiagvik 219 (formerly Barrow), Alaska at 71.3230 degrees north and 256.6114 degrees West (Vasel et al., 220 2020). Continuous atmospheric measurements of CO<sub>2</sub> have been recorded at this observatory since 221 July 1973 (Thoning et al., 2021). Herbert et al. (1986) discuss how the data are processed. 222 223 Peterson et al. (1986) discuss the first ten years (1973-1982) of operations and report consistency of the Barrow results with the reported data from four neighboring locations. Tans and Thoning 224 (2020) provide a general overview of the methods used to collect and process the CO<sub>2</sub> data at 225 Mauna Loa, one of NOAA's other baseline observatories. Along with the hourly temperature data 226 corresponding to BRW, the  $CO_2$  data for BRW were downloaded using the following link: 227 (http://www.esrl.noaa.gov/gmd/dv/data/). 228

229 230

Measurements of downward total solar irradiance have been reported at the BRW observatory since January 1976. Before 1998, the data were reported at three minutes intervals. The data were subsequently reported at one-minute intervals. For this study, the reported values were rolled up to hourly averages. Data were dropped from the analysis if the number of valid minutes of datafor an hour was less than 15.

236

Consideration was given to the inclusion of CH<sub>4</sub> data in the analysis. This action would have 237 resulted in the loss of 26,381 hourly observations due to unavailable or invalid CH<sub>4</sub> measurements. 238 239 (the collection of the CH<sub>4</sub> data commenced in 1986 but was subsequently suspended for about nine months in 2012/2013). The probable effect of this data loss on model convergence was an 240 important consideration in excluding this variable from the analysis, model convergence being one 241 of the major challenges of the methodology employed in this paper (STATA, 2021, p. 33). The 242 omission of CH<sub>4</sub> and other variables reflecting greenhouse gas concentrations represents a 243 shortcoming in the analysis. 244

245

The sample for this study spans from 1 Jan 1985 through 31 Dec 2015. Data before 1 Jan 1985 were not employed in this study because the reported downward total solar irradiance data largely did not meet ESRL's standards before that date. For example, only about 31% of the downward total solar irradiance values in 1984 were deemed by ESRL to be valid. The 1 Jan 2016 - 31 Aug 2017 time interval is reserved for out-of-sample analysis. The evaluation period terminates on 31 Aug 2017 because of a significant data availability issue.

252

In thinking about meteorological issues at BRW, it is useful to begin by first noting the extremes and high level of variability in the level of downward total solar irradiance at this location. In terms of variability, the data from 2014 is instructive (Figure 2). Concerning the

extremes, there are about 67 days of virtually total darkness each year ( about 18 Nov to 22 Jan),





Figure 2. The level of hourly downward total solar irradiance at BRW, 1 Jan 2014 – 31 Dec
2014

261

The average annual temperature at BRW has increased significantly since 1985 (Figure 3). 262 Specifically, the average annual temperature over the 2015-2020 time period was about 3.37 °C 263 higher than in 1985-1990. The temperature data reported by the PABR weather station at the 264 nearby Barrow Airport from 1985 through 2020 are consistent with the trend at BRW (Figure 4). 265 The PABR data also indicates that the four warmest years since 1921 occurred in 2016, 2017, 266 2018, and 2019. In these four years, the average annual temperature was about 5.03 °C higher 267 than the average annual temperature from 1921 through 1939. These findings do not support the 268 assertion by Lindzen that the recent warming is about the same as before the 1940s (2020, pp. 12-269

#### manuscript submitted to AGU Advances

13). In terms of the magnitudes of the recent warming, the increases are consistent with Arctic
amplification, as explained by Pithan & Mauritsen (2014) and Winton (2006).

The upward trend in temperature at both BRW and PABR is consistent with the temperature trend 272 for the Arctic noted by Post et al. (2019), Markon et al. (2018, p 1190-1192), and Thoman et al. 273 (2020, p. 4). Box et al. (2019) have reported significant changes in nine key measures of the Arctic 274 climate system over 1971 through 2017. The qualitative story is clear: "the transformation of the 275 276 Arctic to a warmer, less frozen, and biologically changed region is well underway." (Thoman et al., 2020, p. 1). Consistent with these changes, the annual mean permafrost temperatures have 277 increased at many locations throughout the Arctic (Romanovsky et al., 2017, p. 69). For example, 278 based on data reported by EPA, the average annual permafrost temperature at the Deadhorse 279 Permafrost Observatory (https://permafrost.gi.alaska.edu) over the years 2015 through 2020 280 was about 2.81 °C higher than during the years 1985 through 1990 (EPA, 2021). In four of the 11 281 permafrost observatories whose 2020 annual temperatures are reported by EPA, the 2020 average 282 temperatures were between -1 and 0 °C. There is evidence that thawing has adverse implications 283 284 for carbon emissions because of stimulated microbial decomposition (Schuur et al., 2021).

285

According to AMAP, "Arctic warming can also have effects far beyond the region: for example, the recent rapid warming of the Arctic appears to have created conditions favoring a persistent pattern in the jet stream that provokes unusual extreme temperature events in the Northern Hemisphere." (AMAP, 2019, p. 4). Taylor et al. (2017, p. 303) have indicated it is very likely that human activities have contributed to these trends. While the literature supports this finding, it has also been suggested that the significant natural weather and climate variability in the Arctic poses an attribution challenge (Taylor et al., 2017, p. 319). Consistent with this reported variability, both

downward total solar irradiance and temperature at the hourly level are highly variable (Figures 5 293 and 6). Concerning the hourly CO<sub>2</sub> concentration levels, there is a significant upward trend in the 294 hourly CO<sub>2</sub> concentration levels over the sample (Figure 7). Despite the upward trend in both CO<sub>2</sub> 295 concentrations and temperature, there is no visually obvious relationship between the two variables 296 (Figure 8). While some climate deniers may be tempted to claim that the data in this figure 297 vindicates their position, the view here is that a lack of correlation between two variables only 298 rules out causality when the hypothesized relationship is quite simple. 299



307

- 308
- 309



Figure 4. The average annual temperature at the PABR/Barrow Airport weather station, 1921 2020





12





Figure 6. Hourly downward total solar irradiance levels at the Barrow Observatory, 1 Jan 1985 - 31 Dec 2016



Figure 7. Hourly CO<sub>2</sub> concentration levels at the Barrow Observatory, 1985 -2019



323 324

Figure 8. A scatter diagram of hourly temperature and CO<sub>2</sub> concentration levels at BRW, 1 Jan
 1985 – 31 Dec 2015

The autocorrelative nature of hourly temperature is an important characteristic of the data (Figure 9). As the figure indicates, the magnitude and the duration of the autocorrelative process are significant. In terms of magnitude, the estimated one-hour autocorrelation in temperature equals 0.9970, a value that is so large that it is reasonable to wonder if there is a unit root issue. If this is indeed the case, the results of this study could be spurious for the reasons explained by Kennedy ( 2008, p. 301).

Fortunately, an Augmented Dickey-Fuller test yields a *P*-value that is less than 0.0001 both with and without a possible trend, and thus the null hypothesis of a unit root is rejected. Consistent with this finding, the Phillips-Perron test for a unit root also yields a *P*-value less than 0.0001 both with and without a possible trend. Consideration was given to further unit root testing using the DF-GLS test developed by Elliot et al. (1996). This test is regarded as a leading "secondgeneration" unit root test that avoids some of the shortcomings of the Augmented Dickey-Fuller and Phillips-Perron tests (Baum and Hurn, 2021, pp. 117-120). The application of this
methodology requires a data series without any gaps. The Barrow data set has 325 gaps in terms
of temperature, and thus, the DF-GLS test cannot be applied.

Fortunately, hourly temperature data analysis at another observatory in the polar region 344 may be instructive. One of the few stations in the polar region that substantially meets the zero 345 data gap requirements of the DF-GLS test is the Syowa station on East Ongle Island, located about 346 4km from the Antarctic continent with a latitude 69.0125° South and a longitude of 39.5900° 347 This station is supported by the National Institute of Polar Research in Japan. The data 348 East. from this station was obtained from NASA's CERES/ARM Validation Experiment (https://ceres-349 tool.larc.nasa.gov/ord-tool/jsp/SYN1degEd41Selection.jsp). 350

From 14 Apr 2002 through 31 Jan 2016, a period with 120,982 hours and no data gaps, the mean temperature at the Syowa Observatory was about -10.7 °C, with the hourly values ranging from 41.25 °C to 7.65 °C. At one hour lagged, the autocorrelation in temperature equals 0.9959, a value seemingly suggestive of a unit root issue. This possible suspicion is not supported by the Augmented Dickey-Fuller, Phillips–Perron, or the DF-GLS tests.

356 While the available tests do not support the null hypothesis of a unit root in the hourly

temperature data, a quantitative analysis of hourly time-series temperature data needs to control

its time-series nature to effectively extract the signal from the noise in the data. The method of

ordinary least squares is woefully deficient in this regard. This point is consistent with a warning

- by Granger and Newbold (1974, p. 117), who note the following: "In our opinion the
- 361 econometrician can no longer ignore the time series properties of the variables with which he is

362 concerned - except at his [ or her ] peril." The consequences of ignoring their warning include

363 inefficient estimates of the regression coefficients, suboptimal forecasts, and invalid tests of

364 statistical significance. Unfortunately, an inspection of "Statistical Methods in the Atmospheric

365 Sciences," authored by Wilks (2019), suggests that this warning has not been fully heeded in the



367



The shaded area is the 95% confidence band under the null hypothesis of no autocorrelation

Figure 9. The autocorrelations in hourly temperature at Barrow, 1 Jan 1985 – 31 Dec 2015
370

# 372 5 An ARCH/ARMAX Model of Hourly Temperature

373

371

The model employed in this paper is an Autoregressive Conditional Heteroskedasticity/

Autoregressive–Moving-Average with Exogenous Inputs model of temperature (henceforth, an

376 ARCH/ARMAX model of temperature). The ARCH terms are employed to model the

377 conditional heteroskedasticity, an important consideration in the convergence process. The

378 Autoregressive–Moving-Average (ARMA) component models the autocorrelations in

temperature depicted in Figure 9. In this section, the role of the exogenous inputs is discussed.

Following from Forbes and St. Cyr (2017, 2019) and Forbes and Zampelli (2019, 2020), the 381 modeling approach employed in this paper accepts the proposition that "All models are wrong; 382 some models are useful" (Box et al., 2005, p. 440). They are all "wrong" because they represent 383 a simplification of reality; they can be useful if important features of that reality are captured. A 384 possibly related proposition that may be relevant during these times of sharp differences in 385 opinions is "that all modeling results can easily be dismissed out of hand as being wrong, even if 386 they are useful." In the case of this research, it may be asserted that the results are "wrong" because 387 the model is adversely affected by "specification errors," "multicollinearity," "autocorrelation," 388 "heteroskedasticity," "overfitting," and "unit-root issues." Other readers may conclude that the 389 model is "wrong" because it somehow "forces" the estimated relationship between CO<sub>2</sub> 390 concentrations and temperature to be positive because both are rising over time (note: the 391 correlation between temperature and  $CO_2$  equals -0.1495). Still, others will argue that the results 392 are "biased" because the model's dependent variable is the natural logarithm of temperature. 393

394

Following from Forbes and Zampelli (2020, p. 13), this paper accepts the proposition that the "…vulnerability of a model to be deemed as wrong even though all models are "wrong" represents a challenge to the recognition of insights provided by models that are useful." Fortunately, this challenge can be addressed by assessing a model's predictive accuracy. Common sense informs us that a model that yields accurate predictions is useful if the evaluation interval is sufficiently long. Based on this perspective, the approach in this paper proceeds by estimating the model using 228,085 observations and performing an out-of-sample analysis with 13,175 observations.

402

In the model, the association between  $CO_2$  concentrations and temperature is presumed to be 403 conditional on the level of downward total solar irradiance measured at the Earth's surface, 404 downward total solar irradiance being the primary driver of the weather and climate system. The 405 other drivers of the surface energy balance, such as upward and downward longwave irradiance, 406 are not included as explanatory variables in the model because they are hypothesized to be affected 407 by  $CO_2$  concentrations. Upward short-wave irradiance is not hypothesized to be directly affected 408 by CO<sub>2</sub> concentrations. Its inclusion as an explanatory variable is open to question, given that it 409 is largely driven by downward solar irradiance and temperature. The inclusion of this variable 410 would significantly reduce the sample size, given that ESRL only commenced reporting this 411 variable in 1993. 412

In the model,  $CO_2$  concentrations are lagged one hour to avoid the issue of possible twoway causality between temperature and  $CO_2$  concentrations. The model also includes binary variables representing the solar zenith angle, the hour-of-the-day, day-of-the-year, and year. These variables are included as proxies for the drivers of the diurnal variation in temperature, the seasonal variation in temperature, and the possible non-anthropomorphic drivers of temperature unrelated to total downward solar irradiance. In terms of functional form, linearity is not presumed. Instead, the data are permitted to speak for themselves on this important issue.

420

421 The initial version of the model is given by:

422 InTempt = 
$$\alpha_0 + \alpha_1 \operatorname{ZeroSolart} + \alpha_2 \operatorname{Solart} + \alpha_3 (\operatorname{CO2}_{t-1} \operatorname{*ZeroSolart})$$
  
423  
424 +  $\alpha_4 (\operatorname{CO2}_{t-1} \operatorname{*Solart}) + \alpha_5 \operatorname{Solart} \operatorname{*CO2}_{t-1} + \sum_{h=1}^{9} \beta_h \operatorname{Angle}_h$   
425  
426 +  $\sum_{i=2}^{24} \phi_i \operatorname{HourofDay}_i + \sum_{j=2}^{365} \gamma_j \operatorname{DOY}_j + \sum_{k=1985}^{2014} \delta_k \operatorname{Year}_k$  (1)  
427  
428  
429 Where

430 431	InTempt is the natural logarithm of temperature measured in Kelvin in hour t.
432 433	ZeroSolart is a binary variable. The variable is assigned a value of one if the downward total solar irradiance level at Barrow in period t equals zero. Its value equals zero otherwise.
434	
435 436	Solar <sub>t</sub> equals the downward total solar irradiance level at Barrow in period t.
437	CO2 <sub>t-1</sub> is the atmospheric level of CO <sub>2</sub> concentrations at Barrow in hour t-1.
439 440	PosSolart is a binary variable that equals one if the level of downward total solar irradiance at Barrow in period t is positive. Its value equals zero otherwise.
441 442 443	Angle <sub>h</sub> is a vector of nine variables representing the solar zenith angle.
444 445	HourofDay <sub>i</sub> is a series of 23 variables representing the hour of the day.
446 447	$DOY_j$ is a series of 364 binary variables representing the day of the year.
448 449	Year <sub>k</sub> is a series of 30 binary variables representing the year.
450	Please note that $\alpha_1$ , $\alpha_2$ , and $\alpha_3$ , etc. are the coefficients corresponding to this linear version of the
451	model. From (1), the total number of coefficients to be estimated equals 432. Some may strongly
452	suspect that this number of explanatory variables indicates that the model is "overfitted." If this
453	claim is true, the model would be unlikely to yield accurate out-of-sample predictions even if the
454	within-sample explanatory power is very high (Brooks, 2019, p. 271). The "rule of thumb" by
455	Trout (2006) that overfitting is avoided when there are at least ten observations per estimated
456	coefficient does not support this possible suspicion given that the structural model present in this
457	paper entails over 500 observations per estimated coefficient. Moreover, as will be seen, the model
458	does not suffer from the consequences of overfitting in terms of out-of-sample predictive accuracy.
459	
460	
461	
462	
463	6 Estimation and Desults
464	v Estimation and Results

466	The model was estimated using hourly data over the 1 Jan 1985 - 31 Dec 2015 time interval. The
467	analysis was conducted in two distinct stages. In the first stage, the functional form given by Eq.
468	(1) was evaluated. A nonlinear functional form was subsequentially identified.
469	The analysis also recognizes that the disturbance term's variance in a regression equation is
470	heteroskedastic instead of homoscedastic, i.e., variable instead of constant over time. As
471	suggested in the previous section, the accepted approach involves estimating an ARCH model.
472	This approach was proposed by Engle (1982) to improve the analysis of financial data. It has
473	since proven itself invaluable in modeling any time-series variable in which there are periods of
474	turbulence followed by relative calm at some point. Hourly temperature is one of those
475	variables. Those tempted to claim otherwise are cheerfully invited to consult the book entitled
476	"Environmental Econometrics Using Stata," authored by Baum and Hurn (2021).
477	
478	The second estimation stage also recognizes that the temperature in hour t is not statistically
479	independent from the temperature outcomes in previous hours, as seen in Figure 9. As suggested
480	in the previous section, this is done using an ARMAX specification. In this case, the
481	transformed explanatory variables from the first stage (e.g., $Solar_t^{1/4}$ ) are the exogenous inputs.
482	Given this specification, the disturbance terms are presumed to follow an ARMA specification
483	that models the autocorrelations reported in Figure 9. The ARMA specification applied in this
484	paper is not parsimonious because the autocorrelative process in Figure 9 is not short in duration.
485	It is recognized that this approach runs counter to the traditional time-series philosophy (Box and
486	Jenkins, 1976, p. 17), which suspected that there was more room for prediction errors when more
487	time-series parameters were estimated (Hamilton, 1994, p. 106). The view here is that the goal
488	of predictive accuracy can sometimes be enhanced by including more ARMA terms. This

approach makes sense given the long memory property of the autocorrelations evidenced in 489 Figure 9 and the high level of variability in temperature, as evidenced by Figure 5. The 490 heteroskedasticity is modeled as a function of the solar zenith angle, the hour of the day, the day 491 of the year, the year of the sample, and the following variables:  $\sqrt{CO2_{t-1}}, \sqrt{Solar_t}$ . Instead of 492 assuming that hourly temperature is independent of the conditional variance, the model permits 493 the data to speak for itself on this issue. This linkage is relevant if the level of a variable depends 494 on the variance in the disturbance term. The ARCH-in-mean model introduced by Engel et al. 495 (1987) offers an approach to estimate this linkage. 496

497

The possible merits of representing the explanatory variables using a nonlinear specification are addressed using the multivariable fractional polynomial (MFP) methodology (Royston and Sauerbrei, 2008). Its application includes Forbes and St Cyr (2017, 2019) and Forbes and Zampelli(2019, 2020). The methodology considers the effects of nonlinear transformations of the explanatory variables. In the present case, the MFP results suggest the following specification:

503		
504	$\ln \text{Temp}_{t} = \alpha'_{0} + \alpha'_{1} \text{ZeroSolar}_{t} + \alpha'_{2} \text{Solar}_{t}^{1/4} + \alpha'_{3} (\text{CO2}_{t-1} \times \text{ZeroSolar}_{t})^{3}$	
505		
506	+ $\alpha'_4 (\text{CO2}_{t-1} * \text{PosSolar}_t)^{1/4}$ + $\alpha'_5 (\text{Solar}_t * \text{CO2}_{t-1})^{1/4}$ + $\sum_{h=1}^9 \beta'_h \text{Angle}_h$	
507		
508	+ $\sum_{i=2}^{24} \phi'_i$ HourofDay <sub>i</sub> + $\sum_{j=2}^{365} \gamma'_i$ DOY <sub>j</sub> + $\sum_{k=1985}^{2014} \delta'_k$ Year <sub>k</sub>	(2)
509		
510		

510 511

Please note that  $\alpha'_1$ ,  $\alpha'_2$ , and  $\alpha'_3$  etc. are the estimated coefficients in this specification. Least squares estimation of (2) produces a seemingly respectable level of explanatory power, the R<sup>2</sup> being about 0.831. However, a Portmanteau test for autocorrelation (Box and Pierce, 1970; Ljung and Box, 1978) reveals that the residuals are highly autocorrelated. Consistent with Forbes and St. Cyr (2019, p.17), for lags one through 100, the *P* values are less than 0.0001. The null 517 hypothesis of no ARCH effects is rejected with a *P*-value less than 0.0001. Consistent with these 518 issues, the least-squares model is not useful. This finding is supported by out-of-sample 519 predictions over the period 1 Jan 2016 - 31 Aug 2017 time interval that have a root-mean-squared-520 error (RMSE) of about 5.67  $^{\circ}$  C, a value that is clearly indicative of a suboptimal prediction 521 process.

522

ARCH/ARMAX methods can generate predictions that are much more accurate than the 523 predictions from a least-squares model when the dependent variable is autoregressive and 524 heteroskedastic in nature. In this case, the ARCH process's modeled lag lengths are lags 1 and 2. 525 Consideration was given to including additional ARCH terms to model the apparent diurnal 526 pattern of the ARCH process (e.g., 24, 48, 72, 96 etc.). Consideration was also given to 527 employing alternative ARCH and GARCH specifications. These approaches were abandoned 528 due to model convergence issues. The modeled lag lengths for the AR process are 1 through 12, 529 530 23, 24, 25, 26, 47, 48, 49, 71, 72, 73, 96, 97, 120, 121, 144, 145 167, 168, 169, 192, 193, 216, 240, 264, 288, 312, 335, 336, 337, 360, 384, 408, 432, 456, 480, 600, 671, 672, 673, 840, and 531 960. The MA modeled lag lengths are 1 through 25, 48, 49, 71, 72, 73, 96, 97, 120, 121, 144, 532 533 145, 167, 168, 169, 192, 193, 216, 240, 264, 288, 312, 335, 336, 337, 360, 384, 408, 432, 456, 480, 600, 671, 672, 673, 840, and 960. 534

Equation (2) was estimated assuming that the residual error terms correspond to the Student t distribution instead of the more typical Gaussian distribution. This approach is believed to be justified by the highly volatile nature of the weather system in the vicinity of Barrow. One shortcoming in its application here is that the "degrees of freedom" parameter is less than the minimum indicated by Harvey (2013, p. 20). Consideration was given to modeling the residual

error terms using the generalized error distribution, but this approach was abandoned due to modelconvergence issues.

542

Selected estimates are reported in Table 1. It is revealed that  $\alpha'_2$ , the coefficient corresponding 543 to Solar<sub>t</sub><sup>1/4</sup> is positive and highly statistically significant. The CO<sub>2</sub> coefficients  $\alpha'_3$  and  $\alpha'_4$  are 544 also positive and highly statistically significant while  $\alpha'_5$  is negative and highly statistically 545 significant. These findings are consistent with the view that CO<sub>2</sub> concentrations have implications 546 for hourly temperature but do not address the magnitude. Concerning the possible non-547 anthropomorphic drivers of temperature, it is interesting to note that 16 of the 30 variables in 548 question are statistically significant. With 2015 being represented in the constant term, negative 549 values for a year are consistent with higher predicted temperatures in 2015 than in the year in 550 question. There are 13 such cases. For these cases, the coefficients' median value is -0.00543, a 551 552 value that hardly seems important.

553

The model's explanatory power based on the estimated structural parameters ( all the parameter 554 estimates) is 0.8105 (0.9968.) Those who believe that the latter level of explanatory power is 555 somehow "too outstanding to be true," are cheerfully invited to reinspect Figure 9 and contemplate 556 the concept of autocorrelation and how modeling this autocorrelation can affect a model's level of 557 explanatory power. In any event, the view here follows Hyndman and Athanasopoulos (2018, 558 3.4), who note that true adequacy... " can only be determined by considering how well a model 559 performs on new data that were not used when fitting the model." It is also noted that even though 560 a model's  $R^2$  equivalence is a well-recognized measure of model adequacy, a good case can be 561 made that achieving white noise in the residuals is also important (Becketti, 2013, p. 256; 562

563 Kennedy, 2008, p. 315; and Granger and Newbold, 1974, p. 119). To assess whether this measure

of adequacy is achieved, Portmanteau tests for autocorrelation were conducted for the hourly lags

<sup>565</sup> 1 through 100, 192, 284, and 672. At lag 1, the *P*-value is 0.1958. For the remaining 111 lags that

were assessed, the *P*-values are less than .05, thereby rejecting the null hypothesis of a white noise

567 error structure.

568

569

# 570 **Table 1. Estimation Results**

Variable	Estimated	Absolute	<i>P</i> -Value
	Coefficient	Value of the t-	
		Statistic	
Constant term	-84.5387	3.41	< 0.001
ZeroSolar <sub>t</sub>	0.053421	9.25	< 0.001
Solar <sub>t</sub> <sup>1/4</sup>	0.01102	11.23	< 0.001
$(CO2_{t-1}*ZeroSolar_t)^3$	7.70E-11	7.57	< 0.001
$(CO2_{t-1}*PosSolar_t)^{1/4}$	0.01296	9.04	< 0.001
$(Solar_t * CO2_{t-1})^{1/4}$	-0.00232	10.42	< 0.001
Year <sub>1985</sub>	-0.01111	9.96	< 0.001
Year <sub>1986</sub>	-0.00371	2.36	0.018
Year <sub>1987</sub>	-0.00983	6.91	< 0.001
Year <sub>1988</sub>	-0.00808	6.87	< 0.001
Year <sub>1989</sub>	-0.00498	1.76	0.079
Year <sub>1990</sub>	-0.0033	1.47	0.141
Year <sub>1991</sub>	-0.00285	1.82	0.068
Year <sub>1992</sub>	-0.00664	2.21	0.027
Year <sub>1993</sub>	-0.00265	2.52	0.012
Year <sub>1994</sub>	-0.00339	2.47	0.014
Year <sub>1995</sub>	-0.00384	4.43	< 0.001
Year <sub>1996</sub>	-0.00305	1.73	0.083
Year <sub>1997</sub>	0.001996	1.06	0.288
Year <sub>1998</sub>	0.005733	3.48	0.001
Year <sub>1999</sub>	-0.00766	4.34	< 0.001

Year <sub>2000</sub>	-0.00543	4.26	< 0.001
Year <sub>2001</sub>	-0.00359	2.97	0.003
Year <sub>2002</sub>	0.002124	0.61	0.541
Year <sub>2003</sub>	-0.00658	3.21	0.001
Year <sub>2004</sub>	-0.00449	4.07	< 0.001
Year <sub>2005</sub>	-0.00211	1.11	0.265
Year <sub>2006</sub>	0.000883	0.33	0.743
Year <sub>2007</sub>	0.005622	4.31	< 0.001
Year <sub>2008</sub>	1.92E-06	0	0.999
Year <sub>2009</sub>	0.002597	1.98	0.048
Year <sub>2010</sub>	0.000847	0.38	0.707
Year <sub>2011</sub>	0.001634	0.23	0.81
Year <sub>2012</sub>	-0.00044	0.22	0.829
Year <sub>2013</sub>	0.001147	0.46	0.64.
Year <sub>2014</sub>	0.002601	1.40	0.162
Number of Observations	228,085		
R-Square equivalence based on the full model	0.9968		
R-Square equivalence based on the model's structural component.	0.8105		

Regarding the binary variables not reported above, 336 of the 364 day-of-theyear coefficients are statistically significant, while 22 of the 23 hour-of-theday variables are statistically significant. Only three of the nine solar angle coefficients are statistically significant.

Concerning the AR and MA terms, 44 of the 53 AR terms and 31 of the 61 MA terms are statistically different from zero. Both of the ARCH terms are statistically significant. Only one of the three ARCH-in-Mean terms is statistically significant. Regarding the variables that model the heteroskedasticity in the conditional variance, 298 of the 429 variables are statistically different from zero.

585

# 586 7 The Model's Out-of-Sample Performance

587

The out-of-sample evaluation period consists of 13,175 hours over the 1 Jan 2016 to 31 Aug 2017 588 time interval. Recalling that the dependent variable in the model is the natural logarithm of 589 temperature measured in Kelvin, it might seem that a simple retransformation would yield the 590 optimal predicted value. Unfortunately, merely taking the antilogarithm of the predicted natural 591 logarithm of temperature measured in Kelvin may result in a biased temperature prediction 592 (Granger and Newbold, 1976, pp. 196-197). This bias is easily resolved when the error distribution 593 594 is Gaussian using a method presented by Guerrero (1993). Given the non-Gaussian nature of the error distribution in this case, the matter was resolved by estimating a post-processing regression 595 without a constant term using all of the observations in the sample. The explanatory variable in 596 597 this post-processing regression is the hourly temperature measured in Kelvin, while the explanatory variable in this regression is the antilog of the transformed predicted values. The 598 estimated coefficient corresponding to the explanatory variable equals 0.9999895. The associated 599 R-Square equals 1.0000. The estimated parameter from this regression was used to detransform 600 the out-of-sample transformed predicted temperature values. 601

The out-of-sample predictions were compared with the ERA5 predictions for the same general 602 location. For those unfamiliar with the ERA5 modeling results, it was produced by the Copernicus 603 Climate Change Service at ECMWF. In a significant advance from its earlier databases, it reports 604 hourly values across the globe. The ERA5 hourly temperature values for the Barrow location were 605 https://content.meteoblue.com/en/specifications/dataobtained from Meteoblue 606 ( sources/weather-simulation-data/reanalysis-datasets ). 607

608

The out-of-sample temperature predictions from the ARCH/ARMAX model presented in this 609 paper have a predictive R-square of 0.9962. The predictions are visually more accurate than the 610 ERA5 values for the same general location (Figure 10), although it should be noted that the ERA5 611 values correspond to a grid that includes land and ocean while Barrow represents a land location 612 within that grid. Nevertheless, the ERA5 values may serve as a useful benchmark for the 613 614 ARCH/ARMAX out-of-sample predictions. Regarding the RMSEs, the predictions associated with the ARCH/ARMAX model have an RMSE equal to about 0.682 °C, while the ERA5 615 outcomes have an RMSE of about 3.117 °C. Interestingly, an ordinary least-squares estimation 616 of the ERA5 predictions indicates that the prediction errors are not purely random. Specifically, 617 the prediction error is conditional on the magnitude of the predicted temperature and lagged value 618 of the  $CO_2$  concentration. The latter finding is consistent with the central thesis of this paper. 619 Following Granger's discussion of prediction errors (1986, p. 91), both of these findings suggest 620 a pathway to improving the accuracy of the ERA5 predictions. 621

622

The out-of-sample temperature predictions from the ARCH/ARMAX model are significantly degraded when the estimated effects of  $CO_2$  are ignored (Figure 11). The differential in predictive accuracy is visually apparent if one inspects the vertical distance between the scatter points and the 45° line representing the relationship between predicted and actual temperature when the predictions are perfect. As reported above, the full model presented in this paper has an RMSE equal to 0.682 °C over the evaluation period, constraining the  $CO_2$  estimated effects to be equal to zero results in predictions with an RMSE equal to 3.379 °C.

630

The out-of-sample analysis is supportive of the earlier discussion indicating the unimportance of 631 factors other than CO<sub>2</sub> and the total downward solar irradiance being drivers of the increase in 632 annual temperature over the sample period. Specifically, using the full model, the mean 633 predicted temperature over the evaluation period equals - 8.725218 °C. The mean predicted 634 temperature over the evaluation period is -8.725221 °C if the estimated effects of the binary 635 variables for 1986 through 2014 are constrained to equal zero. In short, the binary variables that 636 control for the possibility of annual temperature being affected by factors other than CO<sub>2</sub> or total 637 downward solar irradiance have virtually no effect on the out-of-sample predicted temperature. 638 Interestingly, the mean actual temperature over the evaluation period equals -8.712713 °C, a 639 very close value to the mean of the predicted values. 640









Figure 11. The ARCH/ARMAX model predictions with and without the  $CO_2$  estimated effects and the actual temperature outcomes, 1 Jan 2016 – 31 Aug 2017.

647

The structural predictions are less accurate than the predictions from the full model but may 648 649 yield useful insights. The predictions from the structural model have an RMSE equal to 5.21 °C while constraining the CO<sub>2</sub> estimated effects to be equal to zero results in predictions with an 650 RMSE equal to 8.29 °C (Figure 12). In short, constraining the estimated effects of CO<sub>2</sub> to be 651 equal to zero reduces the structural model's predictive accuracy. In terms of temperature, the 652 predicted level is significantly lower when the estimated structural effects of CO<sub>2</sub> are ignored 653 (Figure 13). Observe that the difference in the mean levels of predicted temperature is 654 nontrivial. 655









### 669 8 Summary and Conclusion

670

This paper employed an ARCH/ARMAX model with statistical controls for total downward 671 solar irradiance and 426 binary variables to examine the relationship between CO<sub>2</sub> 672 concentrations and hourly temperature at the Barrow Atmospheric Observatory in Alaska. The 673 model was estimated using hourly data over the time interval of 1 Jan 1985 - 31 Dec 2015. The 674 model was evaluated using hourly data from 1 Jan 2016 through 31 Aug 2017. The predictive R-675 square equivalence of 0.9962 over the evaluation period suggests that the model has reduced the 676 677 attribution challenge associated with the significant natural meteorological variability in the Arctic. Consistent with this view, the predictions over the evaluation period are more accurate 678 than the highly regarded ERA5 values for the same general vicinity. Thus, though the model 679 fails to achieve the metric of "white noise" in the standardized residuals, the accuracy of its 680 predictions over the evaluation period indicates that the model is "useful." These results are 681 consistent with the physics that indicates that rising CO<sub>2</sub> concentrations have consequences for 682 temperature, a point that even climate deniers such as Richard Lindzen, William Happer, Roy 683 Spencer, Patrick Michaels, and the other members of the  $CO_2$  Coalition have conceded. What is 684 different is that the model also offers useful insights into the magnitude of the relationship 685 between CO<sub>2</sub> concentrations and hourly temperature. Specifically, the predictions over the 686 evaluation period are significantly more accurate when they reflect the estimated and statistically 687 significant CO<sub>2</sub> coefficients compared to when those coefficients are ignored. The out-of-sample 688 results indicate that CO<sub>2</sub> concentrations have nontrivial implications for hourly temperature. The 689 modeling results also addressed the possible contribution of factors other than CO<sub>2</sub> being drivers 690 691 of increased temperature over the sample. The mean of the out-of-sample predicted temperature

over the evaluation period is not materially affected by these variables, even though some ofthose variables are statistically significant.

694

Given that all models are "wrong," it is a picayune task to dismiss the estimation results reported 695 in Table 1. It is much more challenging to rationally dismiss the implications of the large decline 696 in the out-of-sample predictive accuracy when the estimated  $CO_2$  effects are ignored. One 697 possibility is that some unknown natural factor at work is the true culprit of the decline in 698 predictive accuracy. While climate deniers may find this an attractive explanation for the results 699 presented in this paper, the model's high level of predictive out-of-sample accuracy suggests that 700 701 unknown factors are not an important driver of temperature. There is also the point that attributing the large decline in the out-of-sample predictive accuracy when the estimated CO<sub>2</sub> 702 effects are ignored to an "unknown variable" is highly likely to represent obscurantism as opposed 703 to a conclusion that represents the best of all competing explanations as explained by Lipton (2004, 704 p. 56). In short, the beliefs of the climate change deniers are not supported by the hourly 705 temperature data at NOAA's Barrow Observatory in Alaska. Considering the inadequate results 706 of COP26, this suggests that the current outlook for the Earth's future is quite grim. Research that 707 further illuminates the shortcomings of the views by climate deniers might help matters. One 708 approach being considered is an analysis of the drivers of the hourly surface energy imbalance, a 709 metric that is easily understood as being important but that climate deniers almost never mention. 710 This research path appears feasible using the methods presented here in light of a preliminary 711 analysis indicating that the hourly surface energy imbalance at Barrow and other locations is 712 autoregressive and heteroskedastic. It is not overly optimistic to believe that modeling these 713 properties will facilitate the recognition of CO<sub>2</sub>'s "signal" in the data. 714

715	
716	
717	Acknowledgments
718	
719	The paper's findings are based on the data collected at the Barrow Atmospheric Observatory. I
720	thank the Earth System Research Laboratory (ESRL), Global Monitoring Division (GMD), of the
721	National Oceanic and Atmospheric Administration (NOAA) for supporting the operations of this
722	observatory. I am also grateful to NOAA's National Weather Service and National Centers for
723	Environmental Information for their stewardship of the PABR temperature data. The unit root
724	conclusions rely partly on the hourly temperature data reported by the Syowa observatory in
725	Antarctica. I am thankful to the National Institute of Polar Research in Japan for supporting the
726	operations of this observatory. I am also grateful for NASA's CERES/ARM Validation
727	Experiment (CAVE) stewardship of the data from this observatory. Earlier versions of this paper
728	were presented at the ESIPP/UCD weekly seminar and the ESRI/UCD policy conference. I thank
729	the participants for their feedback. The research was also presented as an e-poster at the 2020
730	AGU Fall meetings in a session entitled, "Convergence Research in Climate Science: How to
731	Move Beyond Disciplinary Silos." I thank the participants of that session for their comments. I
732	thank Jeffery Wooldridge, Nick Cox, and Maarten Buis for sharing their econometric insights. I
733	thank Rick Thoman for sharing his meteorological insights. I also thank Chris St. Cyr and Dina
734	Tady for their candid comments on an earlier draft. Any errors are the full responsibility of the
735	author.
736	
737	
738	ORCID
739	
740	Kevin F. Forbes, <u>https://orcid.org/0000-0002-9521-6845</u>
741	
742	Conflict of Interest
743	The author declares no conflicts of interest relevant to this study.
744	
745	Data Availability Statement
746	
747	Data used in this research and reproducing STATA codes are deposited on Zenodo at
748	<u>10.5281/zenodo.5833580</u> .
749	
750	
751	
752	
753	
754	
755	
756	
757	
758	
759	
760	
761	

762	References
763 764 765	American Association for the Advancement of Science, (2009). 1021Climate Letter, https://www.aaas.org/sites/default/files/1021climate_letter.pdf
766 767 768 769	American Association for the Advancement of Science, (2016). Thirty-One Top Scientific Societies Speak with One Voice on Global Climate Change. Available at <u>https://www.aaas.org/news/thirty-one-top-scientific-societies-speak-one-voice-global-climate- change</u>
770 771 772	AMAP, 2019. AMAP Climate Change Update 2019: An Update to Key Findings of Snow, Water, Ice and Permafrost in the Arctic (SWIPA) 2017. Arctic Monitoring and Assessment Programme (AMAP), Oslo, Norway. 12 pp.
773 774 775	Australian Academy of Sciences et al. (2001) The Science of Climate Change, <i>Science</i> 18 May 2001: Vol. 292, Issue 5520, pp. 1261 DOI: 10.1126/science.292.5520.1261 https://science.sciencemag.org/content/292/5520/1261
776 777 778	Baum, C. F., and Hurn, S. (2021), <i>Environmental Econometrics Using Stata</i> , Stata Press, College Station, Texas
779 780	Becketti, S. (2013). Introduction to time series using stata. College Station, TX: Stata Press.
781 782 782	Box, G.E.P. and Jenkins, G.M. (1976) <i>Time Series Analysis: Forecasting and Control</i> , rev. ed., San Francisco: Holden Day
783 784 785 786	Box, G. E. P., and D. A. Pierce (1970). Distribution of residual autocorrelations in autoregressive-integrated moving average time series models. <i>Journal of the American Statistical Association</i> 65: 1509–1526.
787 788 789	Box, G. E. P., Hunter, J. S. & Hunter, W. G. (2005), <i>Statistics for Experimenters</i> (2nd ed.), John Wiley & Sons.
790	Box, J. E. et al. (2019). Key indicators of Arctic climate change: 1971–2017
791 792	Environ. Res. Lett.14 045010 https://iopscience.iop.org/article/10.1088/1748-9326/aafc1b
793 794 795	Brooks, C., (2019). Introductory Econometrics for Finance, 4th edition, New York: Cambridge University Press
796	CO <sub>2</sub> Coalition. (2015). Carbon Dioxide Benefits the World
797 798	https://co2coalition.org/publications/carbon-dioxide-benefits-the-world-see-for-yourself/
799 800 801	Elliot, G., Rothenberg, T.J. and Stock, J.H. (1996). "Efficient Tests for an Autoregressive Unit Root," <i>Econometrica</i> , 64, 813-836

Engle, R. F. (1982). Autoregressive conditional heteroscedasticity with estimates of the variance 802 of United Kingdom inflation. Econometrica 50: 987-1007. 803 804 Engle, R.F., Lilien, D.M. & Robins, R.P. (1987) Estimating Time Varying Risk Premia in the 805 Term Structure: The Arch-M Model. *Econometrica*, 55, 391-407. 806 https://doi.org/10.2307/1913242 807 808 EPIC (2018). New Poll: Nearly half of Americans are more convinced than they 809 were five years ago that climate change is happening, with extreme weather driving their views. 810 Available at: https://apnorc.org/projects/is-the-public-willing-to-pay-to-help-fix-climate-change/ 811 812 EPA (2021). Climate Change Indicators in the United States. Explanation of the data available 813 at https://www.epa.gov/sites/production/files/2021-04/documents/permafrost td.pdf 814 Raw data available at: https://www.epa.gov/sites/production/files/2021-04/permafrost\_fig-1.csv 815 816 Forbes, K. F., & St. Cyr, O. C. (2019). The Challenge Posed by Space Weather to High-Voltage 817 Electricity Flows: Evidence From Ontario, Canada, and New York State, USA. Space Weather, 818 17. https://doi.org/10.1029/2019SW002231 819 820 821 Forbes, K. F., & St. Cyr, O. C. (2017). The challenge posed by geomagnetic activity to electric power reliability: Evidence from England and Wales. Space Weather, 15(10), 15–1430. 822 https://doi.org/10.1002/2017SW001668 823 824 Forbes, K.F., Zampelli, E.M. (2020). Accuracy of wind energy forecasts in Great Britain and 825 prospects for improvement, Utilities Policy, Volume 67 Page: 101111 826 827 828 Forbes, K.F., Zampelli, E.M., (2019). Wind energy, the price of carbon allowances, and CO2 emissions: evidence from Ireland. Energy Pol. 133. 829 830 Granger, Clive W.J., (1986). Forecasting in Business and Economics, second ed. Economic 831 Theory, Econometrics, and Mathematical Economics. 832 833 834 Granger, C. W. J. & Newbold, P. (1976) Forecasting transformed series, Journal of the Royal Statistical Society, B-38, 189-203. 835 836 837 Granger, C. W. J., & Newbold, P. (1974). Spurious regressions in econometrics. Journal of 838 Econometrics, 2(2), 111–120. https://doi.org/10.1016/0304-4076(74)90034-7 839 840 Guerrero, Victor M. (1993). "Time-series analysis supported by power transformations." Journal of Forecasting 12(1): 37–48. http://dx.doi.org/10.1002/for.3980120104. 841 842 Hamilton, J. D. (1994). Time Series Analysis, Princeton, NJ: Princeton University Press 843 844 Harvey, A. C. (2013). Dynamic models for volatility and heavy tails: With applications to 845 financial and economic time series. New York: Cambridge University Press. 846 https://doi.org/10.1017/CBO9781139540933 847

- Herbert, G.A., Green, E.R., Harris, J.M., Koenig, G.L., Roughton, S.J., and Thaut, K.W., (1986).
- 850 Control and monitoring instrumentation for the continuous measurement of atmospheric CO2
- and meteorological variables, J. Atmos. Oceanic Technol., 3, 414-421.
- 852 DOI: <u>https://doi.org/10.1175/1520-0426(1986)003<0414:CAMIFT>2.0.CO;2</u>
- 853
- Hyndman, R.J., Athanasopoulos, G., 2018. Forecasting: Principles and Practice, second ed.
- 855 OTexts, Melbourne, Australia. OTexts.com/fpp2.
- 856
- <sup>857</sup> IPCC, (2013). Climate Change 2013: The Physical Science Basis. Contribution of Working
- Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker, T.F., Qin, D., Plattner, G.-K., Tignor, M., Allen, S.K., Boschung, J., Nauels, A.,
- Xia, Y., Bex V., & Midgley P.M., (eds.)]. Cambridge University Press, Cambridge, United
- 861 Kingdom and New York, NY, USA.
- 862

- Leahy, P., (2019). Irish Times poll: Climate change' most serious issue' for majority of voters,
- 866 Irish Times <u>https://www.irishtimes.com/news/politics/irish-times-poll-climate-change-most-</u>
- 867 <u>serious-issue-for-majority-of-voters-1.4051713</u>
- Lindzen, R., (2020). On Climate Sensitivity, Available at:
- 869 <u>https://co2coalition.org/publications/on-climate-sensitivity/</u>
- Lipton, P., 2004. Inference to the Best Explanation. London Routledge.
- 871
- Ljung, G.M., Box, G.E.P., 1978. On a measure of lack of fit in time series models. *Biometrika* 65, 297–303
- 874
- Markon, C., Gray, S., Berman, M., Eerkes-Medrano, L., Hennessy, T., Huntington, H., Littell,
- J., McCammon, M., Thoman, R., & Trainor, S. (2018) Alaska. In Impacts, Risks, and
- Adaptation in the United States: Fourth National Climate Assessment, Volume II [Reidmiller,
- D.R., Avery, C.W., Easterling, D.R., Kunkel, K.E., Lewis, K.L.M., May-cock, T.K. & Stewart
- B.C. (eds.)]. U.S. Global Change Research Program, Washington, DC, USA, pp. 1185–1241.
- doi: 10.7930/NCA4.2018.CH26 Available at: <a href="https://nca2018.globalchange.gov/chapter/alaska">https://nca2018.globalchange.gov/chapter/alaska</a>
   881
- National Academies of Science (2005). Joint science academies' statement: Global response toclimate change
- 884 <u>https://sites.nationalacademies.org/cs/groups/internationalsite/documents/webpage/international</u>
   885 <u>080877.pdf</u>
- 886
- 887 Peterson, J.T., Komhyr, W.D., Waterman, L.S., Gammon, R.H., Thoning, K.W., and Conway,
- T.J. (1986). Atmospheric CO2 variations at Barrow, Alaska, 1973-1982, J. Atmos. Chem., 4,
- 889 491-510. DOI:<u>10.1007/BF00053848</u>
- 890

<sup>Kennedy, P. (2008). A Guide to Econometrics, sixth edition, Malden Massachusetts: Blackwell
Publishing</sup> 

- Pithan, F., & Mauritsen, T. (2014). Arctic amplification dominated by temperature feedbacks in
- contemporary climate models. *Nature Geoscience*, 7(3), 181–184.
- 893 <u>https://doi.org/10.1038/ngeo2071</u>
- 894 895
- Post, E., Alley, R. B., Christensen, T. R., Macias-Fauria, M., Forbes, B. C., Gooseff, M. N.,
- Iler, A., Kerby, J. T., Laidre, K. L., Mann, M. E., Olofsson, J., Stroeve, J. C., Ulmer, F.,
- Virginia, R. A., & Wang, M. (2019). The polar regions in a 2°C warmer world. *Sci. Adv.* 5,
   eaaw9883 DOI: 10.1126/sciady.aaw9883
- 900 STATA, 2021. Autoregressive conditional heteroskedasticity (ARCH) family of estimators, in
- 901Time-Series Reference Manual 17 <a href="https://www.stata.com/manuals/ts.pdf">https://www.stata.com/manuals/ts.pdf</a>
- Romanovsky, V.E., Isaksen, K., Drozdov, D.S., Anisimov, O., Instanes, A., Leibman, M.,
- Mcguire A. D., Shiklomanov, N., Smith, S., & Walker D. (2017) Changing permafrost and its
- <sup>904</sup> impacts. In: Snow, Water, Ice and Permafrost in the Arctic (SWIPA) 2017. pp. 65-102. Arctic
- 905 Monitoring and Assessment Programme (AMAP), Oslo, Norway.
- 906 <u>https://www.amap.no/documents/download/2987/inline</u>
- 907
- Royston, P., & Sauerbrei, W. (2008). Wiley series in probability and statistics. In *Multivariable* model-building: A pragmatic approach to regression analysis based on fractional polynomials
- 910 for modelling continuous variables. Chichester, UK: John Wiley.
- 911 <u>https://doi.org/10.1002/9780470770771.scard</u>
- 912 913
- Schuur, E. A. G., Bracho, R., Celis, G., Belshe, E. F., Ebert, C., Ledman, J., et al. (2021).
- Tundra underlain by thawing permafrost persistently emits carbon to the atmosphere over
- 15 years of measurements. Journal of Geophysical Research: Biogeosciences, 126,
- 917 e2020JG006044. <u>https://doi.org/10.1029/2020JG006044</u>
- 918
- Tans, P. and Thoning K, (2020). How we measure background CO2 levels on Mauna Loa.
- 920 NOAA ESRL Global Monitoring Division. Available at:
- 921 <u>https://gml.noaa.gov/ccgg/about/co2\_measurements.html</u>
- 922
- 723 Taylor, P.C., Maslowski, W., Perlwitz, J., & Wuebbles, D. J., (2017). Arctic changes and
- their effects on Alaska and the rest of the United States. In: Climate Science Special Report:
- Fourth National Climate Assessment, Volume I [Wuebbles, D.J., Fahey, D.W., Hibbard, K.A.,
- Dokken, D.J., Stewart, B.C. & Maycock, T.K. (eds.)]. U.S. Global Change Research
- 927 Program, Washington, DC, USA, pp. 303-332, doi: 10.7930/J00863GK
- 928
- Thoman, R. L., J. Richter-Menge, and M. L. Druckenmiller, Eds., (2020). Arctic Report Card
  (2020), <u>https://doi.org/10.25923/mn5p-t549</u>
- 931
- 932 Thoning, K.W., Crotwell, A.M., and Mund J.W. (2021), Atmospheric Carbon Dioxide Dry Air
- 933 Mole Fractions from continuous measurements at Mauna Loa, Hawaii, Barrow, Alaska,

American Samoa and South Pole. 1973-2020, Version 2021-08-09, National Oceanic and 934 Atmospheric Administration (NOAA), Global Monitoring Laboratory (GML), Boulder, 935 Colorado, USA https://doi.org/10.15138/yaf1-bk21 936 FTP path:ftp://aftp.cmdl.noaa.gov/data/greenhouse\_gases/co2/in-situ/surface 937 938 939 Trout, M. D. (2006). Regression, 10k Rule of Thumb for, Encyclopedia of Statistical Sciences, 940 John Wiley & Sons, Inc. 941 942 UNFCCC Secretariat, (2021a). Nationally determined contributions under the Paris Agreement: 943 944 Synthesis Report by the Secretariat https://unfccc.int/sites/default/files/resource/cma2021\_08E.pdf 945 946 947 UNFCCC Secretariat, (2021b). 4 Key Achievements of COP26, https://unfccc.int/news/4-key-948 achievements-of-cop26 949 950 U.S. Department of Commerce, National Oceanic and Atmospheric Administration National 951 Centers for Environmental Information, (2021) Barrow Airport Global Summary of the Year 952 953 1951 - 2020. Available at https://www.ncdc.noaa.gov/cdo-web/search?datasetid=GSOY 954 U.S. Department of Commerce (1951) Local Climatological Data: Annual Summary with 955 Comparative Data, Barrow Alaska 956 Available at: https://www.ncdc.noaa.gov/IPS/lcd/lcd.html? finish=0.9630710752323048 957 958 Vasel, B., Schultz, C., Schnell, R., Stanitski, D., and Thomas, B., (2020). New Arctic 959 Research Facility Opens Door to Science Collaborations. Arctic Report Card: 2020 Update. 960 NOAA Arctic Program. DOI: 10.25923/24rn-c757. Accessed at: https://arctic.noaa.gov/Report-961 Card/Report-Card-2020/ArtMID/7975/ArticleID/908/New-Arctic-Research-Facility-Opens-962 Door-to-Science-Collaborations 963 964 Winton, M. (2006). Amplified Arctic climate change: What does surface albedo feedback have 965 966 to do with it? Geophysical Research Letters, 33(3). https://doi.org/10.1029/2005gl025244 967 Yang, J., Gounaridis, D., Liu, M., Bi, J., & Newell, J. P. (2021). Perceptions of 968 climate change in China: Evidence from surveys of residents in six cities. Earth's 969 970 Future, 9, e2021EF002144. https://doi.org/10.1029/2021EF002144 971 972 Wilks, D. (2019), Statistical Methods in the Atmospheric Sciences, Elsevier, Cambridge, MA. 973 974